CHARGE TRAPPING PROPERTIES OF ALTERNATIVE HIGH-K DIELECTRICS IN MOS DEVICES

By

Xing Zhou

Dissertation

Submitted to the Faculty of the

Graduate School of Vanderbilt University

in partial fulfillment of the requirements

for the degree of

DOCTOR OF PHILOSOPHY

in

Interdisciplinary Materials Science

December, 2006

Nashville, Tennessee

Approved by:

Professor Daniel M. Fleetwood Professor Ronald D. Schrimpf Professor Robert A. Weller Professor Bridget R. Rogers Professor Sokrates T. Pantelides

ACKNOWLEDGMENTS

First I would greatly thank my advisor Dr. Daniel M. Fleetwood for his continuous support in the Ph.D. program. He was always there to listen, to encourage and to provide intensive professional guidance. He showed me different ways to approach the research problem and taught me how to express my ideas and communicate effectively. To complete my Ph.D. would not have been possible without his generous help both in scientific research and life. It has always been my great pleasure knowing him and working under his guidance for my Doctoral degree. His broad and profound knowledge, earnest research attitude and optimistic lifestyle will be engraved on my mind.

I also thank Dr. Ronald Schrimpf and Dr. Sokrates Pantelides with whom I discussed technical details and they also helped me to think through my problems from a different prospective. Dr. Robert Weller has confidence in me and provides insightful comments on my thesis. I thank him for that. Thanks also to Dr. Bridget Rogers for serving on my committee and asking me good questions.

I am also grateful to Leonidas Tsetseris for his theoretical calculations and James A. Felix for helping me set up the experiments and sharing the calculation program; I also would like to thank Evgeni Gusev for supporting this project by providing samples and simulative discussions. Thanks also to Andrew Sternberg for experimental assistance and troubleshooting in the lab.

I am grateful to my fellow graduate students in the Vanderbilt Radiation Effects and Reliability Group, who made the lab a wonderful workplace and home in the past 4 years. Special thanks go to my cutest son Wayhan Z. Jiang for his beautiful smiles everyday and my husband Hairong Jiang for his continuous support and love, and my parents and parents-in-law, for their unselfish support during all the years of my study.

TABLE OF CONTENTS

ACKNOWLEDGMENTS	ii
LIST OF FIGURES	vi
LIST OF TABLES	X
Chapter	
I. INTRODUCTION	1
II. BACKGROUND AND LITERATURE REVIEW	7
Total Dose Radiation Effects and Calculation of Defect Densities Review of Negative Bias Temperature Instability	7 12
III EXPERIMENTAL DETAILS	17
System Setup Devices	17 18
IV. BIAS TEMPERATURE INSTABILITIES: MEASUREMENTS AND THEO	RY 20
Experimental Data Theoretical Calculations Other experimental BTI scenarios-all consistent with current model	20 23 28
V. RADIATION RESPONSE OF ALTERNATIVE HIGH-K DIELECTRICS	30
HfO ₂ -based Devices Al ₂ O ₃ -based Devices	30
VI. COMBINED EFFECTS OF IRRADIATION AND BIAS TEMPERATURE INSTABILITY (BTI)	35
HfO ₂ -based Devices Al ₂ O ₃ -based Devices	35
VII. CHARGE TRAPPING PROPERTIES DURING SWITCHED-BIAS ANNEALING	46
Post-irradiation Annealing	47
Post-constant voltage stress (CVS) Annealing	54
Physical Mechanisms for Charge Trapping Reversibility	59
Contributions from Border Traps	64

VIII. CONCLUSIONS	
APPENDIX	69
A Control code #1: For bias temperature stressing B Control code #2: For C-V measurements:	69 72
REFERENCES	74

LIST OF FIGURES

Figure	Page
1.	Energy band diagram showing ionizing radiation induced charge generation and trap creation in a metal gated, <i>p</i> -substrate MOS capacitor, which was irradiated under positive gate bias. After [45]8
2.	C-V curves for an n-substrate capacitor irradiated to 1Mrad (SiO ₂) with 10 V on the gate. After [50]10
3.	Nomenclature that separates terms used to identify (a) defect location from (b) measured electrical response in MOS devices. The line between an oxide trap and a border trap depends on the time scale and bias conditions of the measurements. After [52]
4.	Classical R-D model results in five regions whose time behaviours are govern by the equations in the insert. D_H is the diffusion constant and kf and kr are forward dissociation rate and reverse annealing rate of SiH bonds, respectively. After [73]
5.	Schematic diagram of measurement system
6.	Leakage current vs. gate voltage for MOS capacitors with thermal SiO_2 and SiO_xN_y/HfO_2 gate dielectrics. The dashed line denotes the leakage current level above which significant charging occurs in the dielectric associated with mechanisms other than NBTI. After [89]21
7.	Oxide- and interface-trap charge densities as a function of stressing temperature for (a) SiO_2 capacitors stressed for 20 minutes at -1.54 MV/cm; and (b) SiO_xN_y /HfO ₂ capacitors stressed for 20 minutes at -1.15 MV/cm. New devices were selected for each stress condition. After [89]22
8.	NBTI steps: (1a-b) released of H from P-H complexes in the depletion region (easy) or deep in Si (infrequent), (2) depassivation, or (3) entry of H ⁺ in SiO ₂ . After [74]
9.	(a) ΔV_{ot} and (b) ΔV_{it} as a function of dose for Al/HfO ₂ +SiO _x N _y /Si pMOS capacitors irradiated with 10-keV X-rays at a dose rate of 31 krad (SiO ₂)/min. The gate biases during irradiation are 0.3V (E _{ox} = 2.4 MV/cm), 0V, and - 0.3V (E _{ox} = - 2.1 MV/cm). The effective oxide thickness is 2.1 nm. All calculated electric fields include the appropriate work function differences between the Al and Si. After [92]
10.	(a) ΔV_{ot} and (b) ΔV_{it} as a function of dose for Al/Al ₂ O ₃ +SiO _x N _y /Si <i>n</i> MOS

10. (a) ΔV_{ot} and (b) ΔV_{it} as a function of dose for Al/Al₂O₃+SiO_xN_y/Si *n*MOS capacitors irradiated with 10-keV X-rays at a dose rate of 31 krad(SiO₂)/min. The gate biases applied during irradiation are -0.3 V (E_{ox} = -2.1 MV/cm), 0 V, and 0.3 V. The effective oxide thickness is 3.1 nm. After [92].....34

- 11. (a) ΔV_{ot} and (b) ΔV_{it} induced by positive bias-temperature stressing with or without positive-bias irradiation as a function of temperature for Al/HfO₂+ SiO_xN_y/Si *p*MOS capacitors. The gate bias is 0.3 V (E_{ox}= 2.4 MV/cm) for both irradiation and PBTS. The open squares represent the degradation by PBTS alone. The half open symbols represent the sum of separate 1 Mrad(SiO₂) positive-bias irradiation (Figure 9) and separate PBTS (open squares in Figure 11). The stress time was 600 seconds at each temperature. After [92]......36
- 13. (a) ΔV_{ot} and (b) ΔV_{it} induced by negative-bias irradiation (V_g = -0.3 V), zerobias irradiation, and positive-bias irradiation (V_g = 0.3 V; E_{ox} = 2.4 MV/cm), followed by negative bias-temperature stressing at 1 Mrad (SiO₂) as a function of temperature for Al/HfO₂ + SiO_xN_y/Si *p*MOS capacitors. The bias applied for NBTS is - 0.3 V. The total dose is 1 Mrad (SiO₂), and the stress times are the same as Figure 11 and Figure 12. After [92]......40
- 14. ΔV_{ot} at 1.0 Mrad (SiO₂) for Al/HfO₂+SiO_xN_y/Si *p*MOS capacitors for: 1) negative-bias irradiation and no BTS; 2) zero bias irradiation and no BTS; 3) positive-bias irradiation and no BTS; 4) no irradiation and NBTS; 5) positive-bias irradiation and PBTS; 6) negative-bias irradiation and NBTS; 7) zero-bias irradiation and NBTS; and 8) positive-bias irradiation and NBTS. The bias stressing temperature is 75 °C. Gate biases are ± 0.3 V or 0 V during irradiation, and ± 0.3 V during BTS. After [92]......41

15. MIL-STD 883, Test Method 1019 flow chart. After [135]......42

- 17. ΔV_{ot} and ΔV_{it} for Al/HfO₂+SiO_xN_y/Si *p*MOS capacitors for a series of consecutive switched bias experiments on the same devices: 1) NBTS for 600 s after positive bias irradiation to 1.0 Mrad(SiO₂); 2) a second 600 s NBTS after a 24 hour wait while the device was stored with no bias (floating) at

- Schematic diagram of the gate stack used here, and electron motion during zero or positive-bias irradiation followed by positive or negative BTS. During PBTS, more electrons compensate trapped positive charge; during NBTS, fewer electrons compensate trapped positive charge. After [92]......49

- 22. Induced ΔV_{ot} (solid symbols) and ΔV_{it} (open symbols) for switched-bias annealing either after irradiation (squares) or after constant voltage stress (circles). The gate bias for irradiation is 0.3 V and the dose is 1 Mrad (SiO₂). The gate bias for constant voltage stress is – 0.5 V and the stress time is 1200 s. Both irradiation and constant voltage stress were performed at room temperature. The switched biases are ± 0.3 V at 75 °C, and the stress time was 600 s for each time interval (NBTS, PBTS) for both cases. After [151]......57

- 25. Schematic diagram of the gate stack used here. Electron exchange and proton (H⁺) motion are depicted during negative BTS (a) or positive BTS (b). Mechanism (1) dominates during switched-bias annealing after irradiation while mechanisms (2)-(4) dominate during post-CVS annealing. During NBTS, electrons are pulled back into the substrate after irradiation (mechanism 1(a). H⁺ can either depassivate Si-H bond (mechanism (2a)) or be pulled off the Si-H bond when there is a spatially co-located suboxide bond (O vacancy), Hf atom, or other similar defect nearby that provides a trapping site for the proton, leaving behind a Si dangling bond that functions as an interface trap (mechanism 1(b)). H⁺ also is attracted to and passivates Si dangling bonds near the interface (mechanisms (2b)-(4b)). After [151].......61

LIST OF TABLES

Table	Page
1. Activation Energies for SiO ₂ -based and HfO ₂ -based MOS capacitors	22

CHAPTER I

INTRODUCTION

Bias temperature instability (BTI) is a degradation phenomenon that occurs mainly in MOS devices and integrated circuits (ICs) [1]-[4]. This instability is seen as an increase both in interface-trap density and oxide trapped charge density when bias is applied on the gate at elevated temperature and/or for long times [2], [5], which causes shifts in threshold voltage and a decrease in carrier mobility. The degradation of these device parameters can lead to circuit failures, both for analog and digital applications [2], [6]-[7]. BTI was not typically a reliability limiting mechanism for older generation technologies where circuits were designed to survive large threshold voltage shifts and mobility degradation levels. However, for modern MOS devices with ultrathin oxides operating at low voltages, where even small voltage shifts can jeopardize circuit performance and influence the long-term reliability, BTI becomes a major reliability issue [8].

For devices used in space systems, radiation exposure is another reliability problem. Electronics are exposed to various forms of radiation, such as electrons, protons, neutrons, and heavy ions. Total ionizing dose irradiation causes the buildup of oxide and interface trap charge. A lot of work has been done in the radiation effects community to investigate changes in MOS threshold voltage shifts after irradiation [9]. However, up till now, not much work has been devoted to studying the combined effects of irradiation and BTI on the degradation and long-term reliability of MOS devices. These combined effects are addressed in this dissertation.

Another challenging problem for modern complementary metal oxide semiconductor (CMOS) technology comes from the materials prospective. For the past 40 years, device size has decreased and device performance has increased greatly due to size scaling associated with "Moore's Law" [10]-[11]. However, modern CMOS devices are rapidly approaching the intrinsic physical scaling limit for Si/SiO₂. Devices with ultrathin oxides operate at rather large electric fields because the device operation voltage cannot be scaled more aggressively than the device dimensions. For example, devices with oxides thinner than ~4-5 nm exhibit large off-state leakage currents (1 to 10 A/cm²) since carriers can easily tunnel directly between the substrate and gate electrode [12]-[13]. This is a big concern for satellite systems and groundbased mobile electronics where power conservation is important. To reconcile the need for reduced off-state leakage currents in highly scaled devices, IC manufacturers are considering several high-k gate dielectrics to replace SiO2 in the future devices [12]-[16]. The main advantage of alternative high-κ dielectrics is they can have higher dielectric constants which make it possible to manufacture a gate stack that is physically thicker than SiO₂, but which maintains electrostatically similar performance to ultrathin SiO₂ layers. The increased physical thickness significantly reduces the probability of tunnelling across the insulator, and therefore reduces the amount of off-state leakage current [15].

The high- κ materials being currently investigated for integration into future IC technologies include Al₂O₃, HfO₂, ZrO₂, Y₂O₃, TiO₂, and Ta₂O₅ and/or the silicates and aluminates of some of these materials [14]-[24]. The dielectric constant ($\varepsilon_{high-\kappa}$) is in the range of 10-40, approximately 3-10 times higher than that of SiO₂ ($\varepsilon_{SiO_2} = 3.9$) [15]. However, some of them have narrow bandgaps, which will lower the barrier height for tunnelling. Since leakage current increases exponentially with decreasing

film thickness and barrier height [25]-[26], this trade-off between dielectric constant and barrier height will determine the relative advantages that an alternative dielectric offers in terms of reduced leakage current. Initial data in the literature show that the net effect for high- κ materials is a reduced leakage current; some high- κ gate dielectrics exhibit up to ~5 orders of magnitude less leakage current than electrically equivalent SiO₂ [18], [27]-[28].

The Group IV metal oxide HfO_2 has been studied extensively for future commercial and space electronics [12], [15], [19], [21], [29]-[30] due to its high dielectric constant (~22) [31]-[32], relatively wide bandgap (~5.6 eV), sufficient band offset (> 1.4eV) [33], and thermal compatibility with Si-based processing [34]-[35]. Al_2O_3 is another promising candidate for its large conduction band offset [36] and good compatibility with high temperature CMOS processing [14]. Also, prior work found that Al_2O_3 exhibited good total-dose radiation hardness because it has a significant density of electron traps [37]-[38] to compensate the effects of radiationinduced trapped holes.

Many high- κ gate dielectrics have shown encouraging electrical characteristics as described above; however, it is difficult to establish a high quality direct interface between high- κ materials and the Si substrate. To improve the interface quality, an interfacial layer (usually oxide or oxynitride) typically is introduced [39]. The benefit of the interfacial layer is to take advantage of the natural Si-SiO₂ interface while also incorporating high- κ dielectrics to increase the capacitance and thickness and thereby reduce the direct tunnelling probability. Plus, most high- κ materials are not good oxygen diffusion barriers [16], [40]-[41]. The presence of an interfacial layer can avoid silicon oxidation to some extent and help the devices survive the standard integrated circuit fabrication process flow. Nitrogen atoms usually are introduced into the interfacial layer to give better control of the gate leakage current, modify the interface properties, and to avoid dopant diffusion from the gate into the channel [40], [42]. Thermal annealing is a common practical nitridation process. Excess nitrogen also can provide an additional trapping site in the near-interfacial SiO₂, inhibit the motion of hydrogen, and decrease the energy barrier for some kinds of critical defect formation reactions. These can enhance charge trapping in the near-interface dielectric, to a degree determined by the concentration and distribution of nitrogen in the gate stack [5], [8].

In order to maintain a high capacitance, the thickness of the interfacial layer should be minimized since its dielectric constant (around 4-7, depending on nitrogen concentration) is less than that of the high- κ dielectrics [43]. However, the breakdown strength is also limited by the interfacial layer. For a given gate bias and assuming an initially charge free dielectric interface, D (= ϵ E) is continuous across the interface: $\epsilon_1 E_1 = \epsilon_2 E_2$ (1)

Here ε_1 and ε_2 are the dielectric constants of the interfacial layer and high- κ dielectric layer, respectively, and E_1 and E_2 are the electric fields of each layer. $\varepsilon_1 < \varepsilon_2$, so $E_1 > E_2$. Therefore, electrical breakdown may occur first in the interfacial layer of a typical gate stack, when a large gate voltage is applied. The thickness of the interfacial layer ultimately limits the operation voltage for the device. Interface engineers therefore need to balance between low equivalent-oxide-thickness (EOT) and high breakdown strength when integrating interfacial layers into standard commercial IC processing.

Other concerns for high- κ dielectrics include several orders of magnitude more traps found in the bulk or interface and significantly lower mobility compared with thermal SiO₂ gate devices. The probability of bias-induced charge trapping in high- κ

gate stacks is extremely high due to the large densities of "intrinsic" defects in the materials that are now available [13], [32], [39]. The charging and discharging of these traps can greatly influence the performance of the devices [44]; these can reduce the drive current due to electrostatic interaction with trapped charges, and cause threshold voltage instabilities over device operation time.

Qualification of high- κ devices for space applications may need more understanding of the charge trapping characteristics and long-term reliability of these materials. In this study, bias-temperature instabilities, radiation response, and annealing characteristics of high- κ materials (mainly HfO₂ or Al₂O₃/oxynitride gate dielectrics) were evaluated. A detailed study of the effects of switched-bias annealing on these same kinds of MOS devices following X-ray irradiation or high-field stress was reported as well. The contributions of oxide, interface, and border-trap charge are evaluated. Physical models of charge buildup and annealing are developed to account for the results. These results can provide insight into the fundamental charge trapping properties of high- κ dielectrics and help to predict the long-term radiation response and reliability of these devices. Hardness assurance implications are also discussed based on the worst-case response of devices in a combined radiation and BTS environment.

This dissertation is organized as follows. Chapter II introduces the basic concepts of different radiation-induced defects and reviews previous work on negative bias temperature stability (NBTI). Chapters III through VII present experimental results and discussion. In particular, chapter III describes experimental details and chapters IV and V present degradation due to negative bias temperature stress and irradiation separately. Then chapter VI summarizes the combined effects of these two.

Further discussion of mechanisms and detailed switched-bias experiments are interpreted in Chapter VII. Chapter VIII concludes the work.

CHAPTER II

BACKGROUND AND LITERATURE REVIEW

This chapter provides background information about radiation-induced defects including oxide, interface and border traps. The techniques used in this research to separate the threshold voltage shifts due to each type of defect are explained in detail. Furthermore, the negative bias temperature instability (NBTI) phenomenon, previous theories and models, and affecting factors are described in sequence.

Total Dose Radiation Effects and Calculation of Defect Densities

The primary effects of total ionizing dose on semiconductor devices include charge accumulation in the oxide and interface trap creation at the Si/oxide interface. Ionizing radiation creates electron-hole pairs (EHPs) in the gate and isolation dielectrics. Some of the radiation-induced charge recombines and does not affect the device performance. The EHPs that survive initial recombination (i.e., the charge yield) will be separated by the oxide electric field, as shown in Fig. 1 [45]. Electrons have a relatively high mobility in SiO₂, so they are swept to the gate in a picosecond or less at typical device operating conditions. However, holes transport slowly toward the Si/SiO₂ interface via defect sites in the oxide [45]. Some of the holes will recombine with the electrons injected from the silicon, and other holes will become trapped in the oxide, forming positive oxide trapped charges. These oxide trapped charges will cause a shift in the threshold voltage and an increase of the leakage current in integrated circuits.

Meanwhile, as holes transport to the interface, protons (hydrogen ions) can be released and react with Si-H bonds at the Si/SiO₂ interface to form interface traps. Those interface traps will lead to a change of threshold voltage and a decrease of carrier mobility, etc.



Figure 1 Energy band diagram showing ionizing radiation induced charge generation and trap creation in a metal gated, p-substrate MOS capacitor, which was irradiated under positive gate bias. After [45].

Although Fig. 1 only shows hole trapping in the oxide, most gate dielectrics can also trap a significant density of electrons. Some high- κ films have exhibited more electron trapping than hole trapping after exposure to ionizing radiation [30], [38]. The radiation-induced oxide trapped charge in SiO₂ and alternative dielectrics is generally net positive [12]-[13], [32], [36], [46]-[47].

Calculations of oxide trapped charge densities and interface trap densities typically are based on the assumption that the interface traps are charge neutral when the Fermi level is at midgap [48]-[49]. In this case, the midgap voltage shift (ΔV_{mg}) is due entirely to trapped oxide charge. The oxide trapped charges shifts the C-V

characteristics; however, the interface traps change the slope. For instance, the "stretchout" of a C-V curve is due to interface traps (ΔV_{it}) . The interface traps are amphoteric. In the upper half of the silicon band gap (above midgap) they are acceptor-like while in the lower half of the silicon band gap (below midgap) they are donor-like. Interface traps usually make threshold voltage (Vt) more positive in nchannel MOSFETs and more negative in p-channel MOSFETs. Positive oxide-trap charge shifts Vt in the negative direction for both n- and p- channel MOSFETs. [50]. Note that this makes negative bias temperature instability (NBTI) a more serious reliability issue for *p*MOSFETs (on *n*-substrates) since the contributions of both oxide and interface traps to the negative threshold voltage shifts make the devices harder to turn "on". For *n*MOSFETs (on *p*-substrates), there is at least partial offset of the effects of interface and oxide traps on threshold voltage. Unless interface trap densities vary significantly across the bandgap, subthreshold measurements of MOSFETs will provide approximately equivalent information as C-V measurements of capacitors when both of them are used to extract oxide trapped charge and interface trap densities.

The changes in interface trap ΔN_{it} (midgap to flatband) and oxide trap ΔN_{ot} charge densities were determined for MOS capacitors by the following equations via the midgap charge separation method [12], [50]:

$$\Delta N_{ot} = -C_{ox} \frac{\Delta V_{mg}}{qA} \tag{2}$$

$$\Delta N_{it} = C_{ox} \frac{(\Delta V_{fb} - \Delta V_{mg})}{qA}$$
(3)

Here C_{ox} is the oxide capacitance, -q is the electronic charge, A is the area, ΔV_{mg} is the midgap voltage shift, and ΔV_{fb} is the flatband voltage shift. Similarly, the voltage shifts due to net-oxide trap charge ΔV_{ot} and interface trap charge ΔV_{it} can be estimated as:

$$\Delta V_{ot} = \Delta V_{mg} \tag{4}$$

$$\Delta V_{it} = \Delta V_{th} - \Delta V_{mg} \tag{5}$$

Please note that all the changes here are the differences between pre- and poststress capacitance-voltage characteristics.



Figure 2 C-V curves for a n-substrate capacitor irradiated to 1 Mrad (SiO₂) with 10 V on the gate. After [50].

The concept of "border traps" was first proposed by D. M. Fleetwood in 1991 [51]. Border traps are oxide traps that are able to exchange charge with the Si on the timescale of the electrical measurements. The ability to exchange charge with the substrate during the measurement makes border traps looks like interface traps electrically; however, the location of these defects are in the oxide, instead of at the interface (Fig. 3) [52]. The microstructure of border traps is still under debate, with no

single defect likely responsible for all border-trap effects in all materials and devices. In some devices, border traps are apparently associated with hydrogen-related defects in the near-interfacial SiO₂ [53]-[54], or other defects not directly associated with trapped holes [55]-[56].Since border traps can sometimes be mistaken for interface traps electronically; separation of the contribution of border traps from real interface traps is performed in this study. The effective border trap density, ΔN_{bt} is obtained by measuring C-V hysteresis and integrating the absolute value of the capacitance difference [57]-[58]:

$$\Delta N_{bt} \approx (1/qA) \int \left| (C_{reverse} - C_{forward}) \right| dV \tag{6}$$

$$\Delta V_{bt} = \Delta N_{bt} \left(q \,/ \, C_{ox} \right) \tag{7}$$



Figure 3 Nomenclature that separates terms used to identify (a) defect location from (b) measured electrical response in MOS devices. The line between an oxide trap and a border trap depends on the time scale and bias conditions of the measurements. After [52].

Review of Negative Bias Temperature Instability

Negative bias-temperature instabilities (NBTI) are associated with the generation of oxide and interface trap charge at the Si/dielectric interface, when negative bias is applied to the gate for long times and/or elevated temperatures [59]. Typical stress temperatures lie in the range of 50 °C-300 °C with oxide electric fields below 6 MV/cm [2]. Fields above this can lead to F-N degradation which is beyond the scope of the investigations in this thesis. NBTI is a long-term reliability issue for modern MOS devices with low operation voltage [60]. Although NBTI has been known for more than 30 years [61], the origins of NBTI are still under debate. In older generations of technology with thick oxides, NBTI often has been associated with water [3], [62], and/or oxygen vacancies near the Si-dielectric interface [63]. For thinner oxides, hydrogen reactions offer a more likely explanation [4], [60].

Among the numerous discussions of NBTI mechanisms, the reaction-diffusion model is the most popular one. The model is described in two steps, i.e., a reaction to create oxide and interface traps and the resulting diffusion and potential subsequent reactions of the initial reaction products. The following equations only provide one of the possible reactions:

(Electrically inactive Si/SiO₂ interface defect) + hole \leftrightarrow (oxide positive charge) + (interface trap) + X_{interface} (8)

$$X_{\text{interface}} \xleftarrow{\text{diffusion}} X_{\text{bulk}} \tag{9}$$

Here X represents a hydrogen-related mobile species that diffuses away from the interface. Hydrogen has long been thought to be critical in NBTI and other defect formation processes in MOS gate dielectrics [2], [5], [59], [61], [64]. For thicker oxides, ΔN_{it} is governed by H₂ diffusion and

$$\Delta N_{it} = A E_{ox}^{m} t^{n} \exp(\frac{-E_{a}}{k_{B}T})$$
(10)

For thinner oxides, the devices tend to show instead a temperature-activated, Eoxindependent long-time ΔN_{it} increase [59], [65]-[71]. The net activation energy can be expressed as a function of dissociation and annealing of Si-H bonds and the diffusion of hydrogen through the oxide [72]. Other than the electric field, temperature, thickness of oxide, and stress time are other important factors that need to be considered for developing an accurate extrapolation model for NBTI degradation to predict device lifetime. Jeppson and Svensson et al [71] proposed first a diffusion controlled mechanism to explain the time dependence of interface trap creation. Others suggested a reaction-limited mechanism [2], [4], [72]. Their assumption is that, as the reaction-limited time dependence obeys a linear relationship. The observed power-law dependence on the time (tⁿ) originates from the limitations on the reaction rates due to the subsequent reactions of the diffusing by-products (hydrogenrelated) of the original reactions. M. A. Alam et al. proposed an "all-purpose" reaction-diffusion (R-D) model to explain the power law dependence--tⁿ observed in the experiments. They believe that interface trap creation is limited by a reaction (i.e., breaking of Si-H bond) at early stage. The subsequent diffusion of hydrogenated species dominates the process afterwards. The power law of time dependence applies over five interface trap generation regimes (shown in Fig. 4) [73]: (1) N_{it} increases due to Si-H bond breaking - here the system is reaction-limited with a characteristic slope of 1 (n = 1); (2) diffusion of hydrogen begins to take over (n = 0) - here the reaction is in equilibrium but the flux of hydrogen away from the interface is negligible; (3) trap creation is limited by hydrogen diffusion (n = 1/4), which is independent of the oxide field and/or the temperature but is only determined by the nature of the diffusing hydrogenated species; (4) hydrogen diffuses in the gate with infinite diffusion velocity (n = 1/2), and (5) the maximum trap density, N₀, is reached due to the saturation of the process, and N_{it} no longer increases with time (n = 0). This is rarely observed in the experiments.



Figure 4 Classical R-D model results in five regions whose time behaviours are governed by the equations in the insert. D_H is the diffusion constant and k_f and k_r are the forward dissociation rate and reverse annealing rate of SiH bonds, respectively. After [73].

The equation in Fig. 4 can be obtained based on the diffusion control model with modified boundary conditions and Fourier transformation. Although the R-D concept is feasible, first-principle calculations based on density functional theory [74] show that direct removal of H is not energetically feasible. More details will be shown in Chapter IV; our revised R-D model will be introduced at the same time.

Empirical models are directly from experiments when NBTI degradation saturates. All the parameters are extracted from the data and the dependence of electric field (E_{ox}) on threshold voltage shifts (ΔV_{th}) is expressed as an exponential function [75]-[76]:

$$\Delta V_{th} = C \exp(\beta E_{ox}) \exp(-\gamma time^{-1/\alpha})$$
(11)

C. E. Blat and E. H. Nicollian proposed a totally different model based on a first-order electrochemical reaction in the form of [2], [77]:

 $\equiv Si-H + A + p^{+} \leftrightarrow \equiv Si \cdot + B^{+}$ (12)

Here A is a water-related species. Induced interface and oxide-fixed charge densities are equal and the rate of defect creation depends on the hole (p^+) concentration and electric field. Water must be present in the oxide near the Si-SiO₂ interface to initiate the reaction. They also believe that the interface trap density equals to the oxide trapped charge density which contradict with the experimental results.

Charge trapping can also be the cause of true NBTI or "NBTI-like" degradation [78]-[83]. The threshold voltage shifts are either due to the tunnelling of holes or a kinetic process initiated by hot electrons injected from the gate electrode. The latter is especially important for higher electric fields than we consider in this thesis, but often are considered NBTI in the literature. Charge transport and trapping lead to oxide and interface trap charge buildup and device degradation.

NBTI models I described so far focus on the degradation of device performance; there is a need for further study on defect formation at atomic level. J. M. Soon et al [84] suggest that released H⁺ will attach to bridging oxygen atoms (Si-O-Si) near the interface. The NBTI-induced defect will weaken the bond strength of Si-O and trigger more breakdown events in the device. J. Ushio calculates the total energies of representative H positions in neutral and positive states where oxygen vacancies serve as hole trapping centers. The results show that the H migration is much easier to occur in the hole-trapping state than in the neutral state, and the activation energy change by electric field is too small to explain the experimentally observed electric-field dependence [85]. L. Tsetseris et al. (our group) suggest that the depassivation of Si-H bonds by protons can account for the generation of interface traps. A possible source of hydrogen is from dopant sites in the depletion region of the Si-substrate [74]. More details will be discussed in Chapter IV. Although no single theory can fully explain all NBTI phenomena, such calculations provide insightful implications for the future study.

Many factors can impact NBTI sensitivity, such as chemical species (hydrogen, nitrogen, water, fluorine, and boron) which were introduced into the devices during the process, orientation of Si-substrate, temperature, the processing of the oxide layer [5], etc. Most of these effects either modify the reaction dynamics or influence the rate at which NBTI occurs within a given process and for a given device geometry. To minimize NBTI degradation, the following methods need to be considered: (1) Improve the quality of the gate oxide and interface. The passivation of traps or dangling bonds with a species more resistant to chemically or charge-induced bond breaking by hydrogen is a possible solution to minimize NBTI degradation. (2) Reduce concentration of certain chemical species. Nitrogen at the interfacial layers and water in the bulk can lower the NBTI activation energy and should be avoided or (in the case of nitrogen) used as sparingly as possible, especially at the Si/oxide interface region. (3) Minimize mechanical stress or strain at the interface. More serious NBTI degradations occur in the strained Si devices. Methods include modifying gate oxide formation temperatures and growth conditions, changes to postoxidation annealing, and/or changing the parameters for ion implantation in the nitride layers or stress compensating layers [86]. (4) Substitute deuterium for hydrogen. Since D_2 has higher binding energies and diffuses slower than hydrogen [87]-[88]. Full understanding of the impact of these issues on NBTI can lead to more easily qualified and stable devices.

CHAPTER III

EXPERIMENTAL DETAILS

System Setup

For the bias temperature-stress experiments described in this thesis, the samples were heated with a hot chuck (Model S-1060R, Signatone) and held at a constant temperature for a period of time under bias. A HP 4140B picoammeter/DC Voltage Source served as bias source. The stress time and temperature can be controlled by program #1 in the appendix A. The changes in interface trap and oxide trap charge densities were determined from high-frequency (1 MHz) capacitance-voltage (C-V) measurements via the midgap charge separation technique described in Chapter II [50]. C-V measurements were executed on a HP 4175A Multi-frequency LCR Meter. The source control code (#2) is shown in the appendix as well. Fig. 5 shows a schematic diagram of the whole measurement process.

Irradiations were performed using an ARACOR Model 4100 10-keV X-ray irradiator. The choices of specific dose rate during irradiation do not significantly affect the results presented in this dissertation. The dose rate used in this study is 31 krad(SiO₂)/min. Different total dose levels were achieved by varying the irradiation time. The typical total ionization doses applied are 500 krad(SiO₂), 1 Mrad(SiO₂), and 2 Mrad (SiO₂).



Figure 5 Schematic diagram of measurement system.

Devices

For NBTI experiments, two types of capacitors were compared. One has a thermal SiO₂ gate oxide; the other has an HfO₂ gate dielectric with a thin oxynitride layer at the Si interface. The standard thermal oxide was prepared by rapid thermal processing at 1000°C in dry O₂ to form a 6.5 nm thick SiO₂ dielectric layer on p-type Si (100) wafers. No post-oxidation annealing was performed. The SiO_xN_y/HfO₂ capacitors were fabricated on HF-last p-type Si (100) wafers with a doping concentration of ~10¹⁶ cm⁻³ [89]. The physical thickness of the high- κ layer was 6.8 nm as measured ellipsometrically; the interfacial oxynitride layer was 1.0 nm. The interfacial oxynitride films were thermally grown in a mixture of NO and O₂ at 900°C, resulting in ~ 10% to 15% N incorporation. The high- κ layers were deposited by atomic layer deposition (ALD) at 300°C using conventional surface chemistries.

The relative dielectric constants (ϵ_r) of the HfO₂ and interfacial oxynitride layer (SiO_xN_y) are ~ 20 and ~ 4, resulting in an equivalent oxide thickness (EOT) of 2.1 nm. Al gates were deposited to form gate electrodes. Atomic layer deposition (ALD) is the most promising high- κ deposition technology [15], [90]-[91]. ALD films are formed by repeating a sequence of alternating surface-reactant interactions, which are saturating and "self-limiting." This technique gives highly conformal films with monolayer control of film thickness, uniformity, and materials properties.

For the radiation experiments, two types of high- κ based MOS capacitors are compared. One has HfO₂ dielectrics and the other has Al₂O₃ dielectrics. The HfO₂ capacitors were formed on p-type (100) Si as described previously, but the Al₂O₃ capacitors were formed on n-type (100) Si. The Al₂O₃ layer was deposited by atomic layer deposition (ALD) at 300°C using standard Al (CH₃) + H₂O surface chemistries [14], [18], [92]-[93]. After deposition, the dielectrics were subjected to a forming gas (5-10% H₂ in N₂) anneal (FGA) at 550°C. 1.0-nm-thick interfacial oxynitride films between Al₂O₃ and the Si substrate were thermally grown in a mixture of NO and O₂ at 900°C, resulting in ~10% to 15% N incorporation. After the annealing, the Al gate electrodes were evaporated at room temperature. The relative dielectric constant of asdeposited Al₂O₃ is ~8 and the relative dielectric constant of silicon oxynitride is ~4-5 for this concentration of N. The equivalent oxide thickness (EOT) of the dielectrics studied here are 3.1 nm.

The combined effects of irradiation and the subsequent BTS were studied on both kinds of MOS capacitors with high- κ dielectrics. Detailed switched-bias experiments either after X-ray irradiation or constant voltage stress (CVS) were performed on Al/HfO₂+SiO_xN_y/Si *p*MOS capacitors.

CHAPTER IV

BIAS TEMPERATURE INSTABILITIES: MEASUREMENTS AND THEORY

In this chapter, experimental results of negative bias temperature stressing degradation are presented first. Experimental activation energies were extracted and compared with theoretical values based on first-principle calculations and revised reaction-diffusion (R-D) theory. The results show that excess hydrogen plays a fundamental role in the defect creation process.

Experimental Data

For the NBTI measurements, the MOS capacitors were heated with a hot chuck and held at a constant temperature under negative bias. The holding temperatures vary from 50 °C to 300 °C and the applied electric fields are between – 1 MV/cm and – 2 MV/cm. The post-stress 1-MHz high-frequency capacitancevoltage curves were taken after the samples cooled to room temperature. Voltage shifts due to net oxide-trap ΔV_{ot} and interface-trap charge ΔV_{it} were estimated via the midgap charge separation method of Winokur et al. [50].

To ensure that the measured NBTI degradation is not affected by charging effects due to high currents through the oxide [94]-[95], the gate current was monitored during all experiments with a HP 4140B picoammeter. This is particularly a concern for the SiO_xN_y/HfO_2 capacitors, as illustrated in Fig. 6. Data sets for which injected charge densities exceeded a threshold of ~ 10^{-4} C/cm² (above which carrier injection and trapping can dominate over NBTI) were not included in this study.



Figure 6 Leakage current vs. gate voltage for MOS capacitors with thermal SiO_2 and SiO_xN_y/HfO_2 gate dielectrics. The dashed line denotes the leakage current level above which significant charging occurs in the dielectric associated with mechanisms other than NBTI. After [89].

Despite this limitation, activation energies for NBTI from Arrhenius plots for an adequate range of temperatures and biases for both device types can still be extracted. A large number of devices were characterized for NBTI over a wide range of bias/temperature stress conditions; results under test conditions that showed maximum, reproducible (to within experimental variation) NBTI with minimum gate leakage are illustrated in Fig. 7. Here Fig. 7 shows ΔN_{ot} and ΔN_{it} for capacitors stressed for 1200 s as a function of temperature at an applied electric field of -1.54 MV/cm for the thermal oxide, and an applied field of -1.15 MV/cm for the SiO_xN_y/HfO₂ oxide. The activation energies (Ea)are listed in Table 1:

These results are comparable to the range of activation energies for NBTI that have been reported in the literature [59], [61], [64], [96].

Table 1 Activation Energies for SiO₂-based and HfO₂-based MOS capacitors

Defect Type	Ea (eV)		
Defect Type	SiO ₂ -based MOS-C	HfO ₂ + SiO _x N _y MOS-C	
ΔN_{ot}	0.27 ± 0.03	0.35 ± 0.04	
ΔN_{it}	0.31 ± 0.04	0.22 ± 0.03	



Figure 7 Oxide- and interface-trap charge densities as a function of stressing temperature for (a) SiO_2 capacitors stressed for 20 minutes at -1.54 MV/cm; and (b) SiO_xN_y/HfO_2 capacitors stressed for 20 minutes at -1.15 MV/cm. New devices were selected for each stress condition. After [89].

Similar activation energies for oxide-trap and interface-trap charge formation due to NBTI in MOS capacitors with SiO₂ and HfO₂+SiO_xN_y gate dielectrics were found. And these activation energies are consistent with lateral motion of protons at the interface. One possible origin for this defect formation is H^+ release from oxygen protrusions in the near-interfacial Si, its diffusion along the Si/dielectric interface, followed by its reaction with Si-H to form interface traps, or its capture at a sub-oxide bond to form positive oxide-trap charge. These oxygen protrusions can release the hydrogen as H⁺ after these complex capture a hole [89], [97]-[100]. The barrier for subsequent lateral motion of H⁺ along the Si/SiO₂ interface is ~ 0.3 eV [101]. At least some of the differences in the levels of defect formation between the SiO₂ and SiO_xN_y/HfO₂ devices in this study may then be explained by differences in the amount of hydrogen in the two types of devices, as well as differences in O vacancy densities in the near-interfacial SiO₂. These results are consistent with the key roles played by hydrogen in MOS defect formation in MOS radiation response and longterm reliability [102], and suggest that minimizing excess hydrogen, O vacancies, and/or oxide protrusions into Si may help to reduce NBTI.

Theoretical Calculations

Although the activation energies in Fig.7 are similar to calculated energies for processes such as lateral proton migration along the Si/SiO₂ interface, it is difficult to understand why such low-energy processes should be observed at or above room temperature. Futher theoretical calculations by L. Tsetseris et al.[74] were performed after the publication of Fig. 7 that are based on density functional theory (DFT) to provide a better understanding of NBTI on atomic scale. The DFT calculations were performed with gradient corrections for the exchange-correlation functional. Plane

waves (cutoff ~ 650 eV) were used as a basis set, utilizing untrasoft pseudopotentials [103] as implemented in the VASP code [104]. Large enough periodic supercells were used to provide realistic geometries and to ensure convergence of results.

The Si-H bond is the most typical precursor for interface traps, as the removal of the H atom leaves behind dangling bonds, which are identified in the literature as P_b centers [48]. The direct dissociation of Si-H bonds at the Si/SiO₂ interface has been suggested as a mechanism for generation of traps during NBTI when holes are present, as discussed earlier in Chapter II.

$$\text{Si-H} \rightarrow \text{Si}^- + \text{H}$$
 (13)

However, DFT calculations show that the removal of the H atom from the Si-H bond to a remote Si-Si bond raises the energy of the system by ~ 1.9 eV. Adding the associated migration barrier, which is ~ 0.5 eV, results in the total dissociation activation energy of 2.4 eV [74], in agreement with experimental measurements (2.6 eV) [105]-[108]. When hole are present, as is the case for NBTI in pMOS transistors, the energy required to remove H decreases to 1.6 eV and E_a decreases to 2.1 eV. These values are well above those expected to affect NBTI, at least in the absence of high electric fields, and/or the existence of additional nearby defects. Hence, the direct dissociation mechanism is not the case for the normal operation of commericalgrade MOS devices.

Given that direct dissociation of Si-H bonds is not the cause for NBTI, the most plausible alternative is the depassivation reaction [74], [98], [100], [109]:

$$Si-H + H^+ \rightarrow D^+ + H_2 \tag{14}$$

First-principles DFT calculations find the reaction energy and barrier for process (14) to be 0.5 eV and 0.95 eV respectively, when the Fermi level is at the valence band maximum. This is for n-type Si under inversion. Hence this process is

energetically preferred to direct dissociation and this process can reach quasiequilibrium for long stress times.

The mechanisms responsible for NBTI for n-type Si are described schematically in Fig. 8. The proposed mechanism for depassivation of dangling bonds (14) requires a source of H. Possible stable and metastable binding sites for H include oxygen protrusions, interstitial O in Si (Si-O-Si configurations) and dopants. These possibilities have been explored, and it is found that, in n-type Si (that is, for pMOS transistors), where H exists as H⁻ (due to its "negative-U" properties in Si [110]), it only binds to dopants. Calculations verify that the equilibrium H position in a p-H complex is the so-called antibonding (AB) site [111]. The energy required to move H from the AB site to a remote position (binding energy Δ_{PH}) in Si is 0.6 eV, combined with the migration barrier 0.7 eV for H⁻ [112], gives an activation energy of 1.3 eV for dissociation of P-H complex, in agreement with the experimental value of 1.18 eV [113]. Hence, excess hydrogen can facilitate the breaking of the Si-H bond much more easily through chemical reaction than can the direct interaction of the holes at the interface.

The stability of the P-H complex changes dramatically when the charge state of hydrogen changes to neutral (H⁰). H stays neutral for a certain period of time after its release in the depletion region under NBTI stress, even though at equilibrium H exists only as H⁺ or H⁻. The binding energy Δ_{PH} drops from 0.6 eV to 0.2 eV for neutral H. Plus the diffusion barrier of H⁰ in Si (0.1- 0.2 eV) [110] is smaller than H⁺ and H⁻ diffusion barriers (0.48 eV and 0.7 eV[112] respectively). The resulting activation energy (E_a⁰) to release H⁰ from a P-H complex is only 0.3-0.4 eV. This value is in agreement with the measured E_a⁰ (0.3 eV) for minority-carrier-enhanced dissociation of P-H complexes in the depletion region of Si [114]. Because of the very small activation energy, all of the H in P-H complexes in the depletion region is released and can migrate rapidly with a small barrier for H^0 (0.1-0.2 eV, shown in Fig.8 (1a)). Also at 100-200°C, some hydrogen can be released from complexes deeper in the Si substrate (illustrated as Fig. 8(1b)). Hydrogen that arrives or is released in the inversion layer becomes positively charged (H^+) by trapping a hole under negative bias; this H^+ is swept to the interface by negative bias. The inversion layer and the depassivation reaction combined act as a "sink" for hydrogen motion in Si. The whole process is depicted in Fig. 8. Because of the relatively high density of dopant atoms in the substrate, only a very small percentage of P atoms must bind a hydrogen atom prior to NBTI in order to produce the numbers of interface traps typically measured in NBTI experiments.



Figure 8 NBTI steps: (1a-b) release of H from P-H complexes in the depletion region (easy) or deep in Si (infrequent), (2) depassivation, or (3) entry of H^+ in SiO₂. After [74].
Protons that reach the interface encounter a large barrier of 1 eV [99] to enter SiO_2 , so they first migrate rapidly along the Si side of interface. The passivation of dangling bonds by transporting protons is suppressed since both of them are of the same charge. So these protons cannot passivate pre-existing interface traps. Additionally, the energy for H⁺ at the center of a Si-Si bond next to a Si-H entity is lower by 0.2 eV with respect to other sites for H⁺ in Si. Therefore, hydrogen can easily find Si-H bonds and activate the reaction (14). At 100-200°C, this reaction reaches quasi-equilibrium quickly and after this point the dynamic balance is controlled by the diffusion of the products H₂. In this diffusion-limited regime, E_a is given by [4], [115]:

$$E_a = \frac{\Delta E}{2} + \frac{\Phi_D}{4} \quad . \tag{15}$$

In Eq (15), ΔE is the reaction energy of process (14), and Φ_D is the diffusion barrier of the migrating species. According to first-principles calculations [74]: $\Delta E =$ 0.5 eV, $\Phi_D = 0.45$ eV [116] for H₂ diffusion in SiO₂, so E_a ~ 0.36 eV which is in very good agreement with previous experimental data [59], [64], [85], [89], [96].

Protons that arrive at the interface can also eventually migrate into the oxide where they contribute to the buildup of oxide trapped charge. Significantly, oxide-trap charge in the form of a proton is much more difficult to neutralized via electron tunneling than trapped holes, leading to the relative stability of trapped H⁺ in SiO₂ under NBTI stress conditions [89], [98]. In the asymptotic limit, the rate of such buildup is controlled by diffusion of the "product" H⁺ in SiO₂. DFT calculations show that the diffusion barrier (Φ_D) of H⁺ in SiO₂ is ~0.8 eV [102], [117], and change in energy as H⁺ enters the oxide (ΔE) is 0-0.2 eV. Substituting these values in equation (15), it is found that E_a = 0.2 - 0.3 eV. For the increase of oxide trapped charge, this is consistent with measured values [64], [85], [89]. As for the formation of interface traps, the small E_a observed in experimental studies is only an apparent activation energy, observed at the large stress time limit, after the migration of H^+ from the Si to SiO₂ reaches quasi-equilibrium.

Other experimental BTI scenarios-all consistent with current model

Other key BTI issues will now be addressed. The effect of negative bias is related to the large reduction of the P-H dissociation energy in the depletion region and the appearance of a boundary condition (sink of hydrogen) that biases the migration of H in the direction of the interface. A positive bias (PBTI) on a *p*-MOSFET can still result in degradation, but in this case the rate limiting step is the thermal dissociation of P-H complexes (barrier 1.18 eV [113]) in the accumulation region. This process may help to explain the results of Zhang and Eccleston [7] and in particular their large extracted E_a (1.23 eV) for PBTI.

Degradation can occur for p-type Si, as indicated in Fig. 8 above, however, so this degradation is typically smaller than for n-type Si. This is because the E_a for dissociation of B-H complexes (1.28 eV [118]) is larger than the P-H value. The release of H is thus more difficult in B-doped Si, leading to a less pronounanced BTI degradation.

For NBTI on p-type Si, the rate limiting step for small time is the thermal dissociation of B-H complexes, consistent with first-order reaction kinetics observed by Blat et al [2], and higher T and/or longer times [59] are required to reach the diffusion-limited regime. However, in this case, another possiblity could be the release of H from O interstitutials in the near interfacial Si region [89]. Such O atoms in the substrate have been shown to exist in certain cases in concentrations large enough to account for NBTI on p-type Si [97].

Any step that can lead to an increase of H content in the substrate, for example by annealing in H₂ or H₂O ambients [2], [7] will result in an increase for NBTI degradation. Other species, such as fluorine [119], or nitrogen in the near interfacial SiO₂ [64], [69], [72], [78], can also affect the kinetics of the depassivatin reaction and impact NBTI degradation. Fluorine has two-side effects on NBTI degration: on the one hand, Si-F interfacial bonds are resistant to depassivation by hydrogen species; on the other hand, an excess of fluorine can create stretched bonds, which can act as carrier traps, or disrupt an abrupt interface by cleavage of Si-Si bonds. More discussions can be found in ref [119].

In summary, NBTI induced oxide trapped charge and interface trap densities were determined at different temperatures. Activation energies were extracted from Arrhenius plots and those results are in good agreement with the therotical calculations based on our revised R-D model. The depassivation of Si-H bonds by protons as a reaction that can account for the observed generation of interface traps. The trapped protons in the oxide attribute to the oxide trapped charge buildup. The possible source of hydrogen needed to initiate depassivation is identified as such species released from dopant sites in the substrate. Finally, other key issues of NBTI are discussed briefly and those discussions are consistent with the present model.

CHAPTER V

RADIATION RESPONSE OF ALTERNATIVE HIGH-K DIELECTRICS

This chapter describes the radiation response of HfO₂-based and Al₂O₃-based MOS devices. Different bias conditions and total dose levels are chosen during irradiation of each group. Effective trapping efficiencies are determined from experimental results.

*HfO*₂-based Devices

Fig. 9 shows results for Al/HfO₂+SiO_xN_y/Si *p*MOS capacitors irradiated with 10-keV X-rays to 2.0 Mrad(SiO₂). The largest shifts in both ΔV_{ot} and ΔV_{it} are observed for positive-bias irradiation, consistent with most (but not all [120]) experience with thermal SiO₂.

Trapping efficiency is a dimensionless quantity used to approximate the intrinsic defect density of an insulator [121]-[122]. Effective trapping efficiencies (f_{ot}) of high- κ dielectrics were defined as what the trapping efficiency would be if the gate dielectrics were SiO₂ [12]:

$$f_{ot} = -\frac{\Delta V_{mg} \varepsilon_{ox}}{q \kappa_g f_y t_{eq} t_{phys} D}$$
(16)



Figure 9 (a) ΔV_{ot} and (b) ΔV_{it} as a function of dose for Al/HfO2+SiOxNy/Si pMOS capacitors irradiated with 10-keV X-rays at a dose rate of 31 krad (SiO2)/min. The gate biases during irradiation are 0.3V (E_{ox} = 2.4 MV/cm), 0V, and - 0.3V (E_{ox} = - 2.1 MV/cm). The effective oxide thickness is 2.1 nm. All calculated electric fields include the appropriate work function differences between the Al and Si. After [92].

Here ΔV_{mg} is the midgap voltage shift, ε_{ox} is dielectric constant of SiO₂ (~ 3.5 × 10⁻¹³ F/cm), -q is the electronic charge, κ_g is the number of electron-hole pairs (EHP) generated per unit dose, f_y is the charge yield, t_{eq} is the equivalent oxide thickness (EOT), t_{phys} is the physical thickness of the alternative dielectric, and D is the total dose [12], [121]. Currently, κ_g and f_y are not known for alternative dielectrics. To first order approximation, it is possible to use the known value for SiO₂ scaled by the ratio of the band-gap of SiO₂ to the band-gap of the high- κ materials since this equation (16) is used to compare with SiO₂. The t_{phys} in equation (16) is for the charge generation throughout the entire volume of the oxide, while the t_{eq} term is to account for the moment arm effect resulting from the spatial distribution of the charges in the oxide projected to the interface [123]. In similar equations for SiO₂, both of these effects are accounted for by a single t_{ox}^2 term [121]. However, for high- κ dielectrics, in order to still use ε_{ox} in equation (16), it is necessary to distinguish between the electrical thickness (EOT) and the physical thickness [12].

Here, $f_y = 0.45$, and $\kappa_g = 1.2 \times 10^{13} \text{ cm}^{-3} \text{rad}^{-1}$. This is the known charge generation value of SiO₂ (~ 8.1 × 10¹² cm⁻³ rad⁻¹) scaled by the ratio of the bandgap difference between SiO₂ and HfO₂. For 1 Mrad(SiO₂) irradiation at positive bias, an effective net oxide-trap charge trapping efficiency (ignoring potential dose enhancement effects from the Hf) is estimated as ~ 22% for these HfO₂ devices, which is similar to other results for high- κ devices in the literature [12]-[13], [46]. However, at the same total dose level, the trapping efficiency of the thermal oxides can range from a few percent up to ~50%, depending on the number of oxygen vacancies in the oxide and oxide quality itself [12], [121]-[122].

Al₂O₃-based Devices

Fig. 10 shows the radiation responses of Al/Al₂O₃+SiO_xN_y/Si *n*MOS capacitors irradiated with 10-keV X-rays to 2.0 Mrad(SiO₂). The largest shifts in both ΔV_{ot} and ΔV_{it} are observed for negative-bias irradiation for these step-stress experiments, in contrast to what was observed in Fig. 9 for the HfO₂ devices. In devices with thicker Al₂O₃ dielectric layers, Felix et al. observed roughly equal values of ΔV_{ot} for positive and negative bias irradiation [13].

Taken together, these results show that more charge trappings occur in the bulk of the insulators for these devices, with the relative densities of positive and negative charge varying with processing conditions and/or dielectric thicknesses. The interface-trap buildup is greatest during negative bias in these devices, and may be caused by hydrogen that is released in the Si during the irradiation [74], [89]. For 1 Mrad(SiO₂) and negative-bias irradiation, the effective net oxide-trap charge efficiency [12] is ~23% for these Al₂O₃ devices, similar to the trapping efficiency of the HfO₂-based devices discussed above.

In summary, the radiation response of MOS capacitors with HfO₂ and Al₂O₃ dielectrics were evaluated and compared with thermal SiO₂. Their calculated effective net oxide-trap charge efficiencies are comparable to values in the literature. For HfO₂-based MOS capacitors, positive bias irradiation gives the largest degradation among all the other bias radiations; however, for Al₂O₃-based MOS capacitors, negative bias irradiation has the largest voltage shifts.



Figure 10 (a) ΔV_{ot} and (b) ΔV_{it} as a function of dose for Al/Al₂O₃+SiO_xN_y/Si *n*MOS capacitors irradiated with 10-keV X-rays at a dose rate of 31 krad(SiO₂)/min. The gate biases applied during irradiation are -0.3 V (E_{ox} = -2.1 MV/cm), 0 V, and 0.3 V. The effective oxide thickness is 3.1 nm. After [92].

CHAPTER VI

COMBINED EFFECTS OF IRRADIATION AND BIAS TEMPERATURE INSTABILITY (BTI)

Detailed investigations of the combined effects of irradiation and bias temperature stress (BTS) for HfO₂-based MOS devices have been performed. The worst-case response of HfO₂ devices in a combined radiation and BTS environment is found to differ significantly from typical worst-case operating conditions for transistors with thermal SiO₂ gate dielectrics. For comparison, high- κ dielectric stacks incorporating Al₂O₃ were also evaluated for their combined irradiation and BTS response. Hardness assurance implications on both types of devices are discussed in this chapter.

*HfO*₂-based Devices

Fig. 11 shows combined effects of positive-bias irradiation and positive BTS (PBTS) on MOS devices with HfO₂ dielectrics. The combined irradiation and BTS values of ΔV_{ot} and ΔV_{it} exceed the sums of the separate irradiation and BTS results. Here the values of both ΔV_{ot} and ΔV_{it} for the combined stresses (solid circles) exceed those for the sums of the separate irradiation (Fig. 9) and pure BTS (open squares in Fig.11), denoted by the dashed curves in Fig.11. The degree to which combined stresses exceed the sums of the individual stresses increases with increasing temperature.



Figure 11 (a) ΔV_{ot} and (b) ΔV_{it} induced by positive bias-temperature stressing with or without positive-bias irradiation as a function of temperature for Al/HfO₂ + SiO_xN_y/Si *p*MOS capacitors. The gate bias is 0.3 V (E_{ox} = 2.4 MV/cm) for both irradiation and PBTS. The open squares represent the degradation by PBTS alone. The half open symbols represent the sum of separate 1 Mrad(SiO₂) positive-bias irradiation (Figure 9) and separate PBTS (open squares in Figure 11). The stress time was 600 seconds at each temperature. After [92].

The small increases in ΔV_{ot} and ΔV_{it} for PBTS without irradiation is likely due to the release and motion of protons within the oxide layers [124]. The enhancement of ΔV_{it} in Fig.11(b) is consistent with two-stage interface-trap buildup during irradiation and positive-bias annealing [125]-[126], but the enhancement of ΔV_{ot} differs from the reduction of net oxide-trap charge usually observed for irradiated SiO₂ with increasing anneal time at elevated temperatures [127]. This increase in ΔV_{ot} in Fig.11(b) likely is associated with radiation-enhanced H⁺ motion and trapping at defect sites near the dielectric to Si interface during PBTS [5], [7], [53], [74], [89], [99], [128]-[129]. The amount of hydrogen in the films and the nature of its transport and trapping are expected to be a strong function of device processing [124].

The enhancement of ΔV_{ot} also differs from the response of Al₂O₃ under similar experimental conditions [38]. Due to high temperature forming gas annealing (FGA) processing, there are more "deep" electron trappings in the Al₂O₃ dielectrics to compensate trapped holes either through thermal annealing or tunnelling [38]. The processing dependence certainly warrants follow-on study and may crucially determine the relative amounts of hole and electron traps in the near-interfacial region of the high- κ dielectrics and impact their post-radiation responses.

In Fig. 12, HfO₂ capacitors are subjected to negative BTS (NBTS) with and without irradiation exposure. Values of ΔV_{ot} and ΔV_{it} with no irradiation are comparable to those observed in previous studies of NBTS in high- κ devices [34], [89], [130]. Values of ΔV_{ot} and ΔV_{it} after combined negative-bias irradiation and NBTS are much larger than the pure irradiation results in Fig. 9 at these doses (magnitudes of ΔV_{ot} and $\Delta V_{it} \sim 20$ mV at 2 Mrad), the pure NBTS results in Fig. 12 (magnitudes of ΔV_{ot} and $\Delta V_{it} \leq 60$ mV at 175 °C), or the sums of the two (dashed

curves). This enhancement in combined response likely is evidence of dipolar charge trapping in the HfO₂ layers, which will be discussed in Chapter VII.



Figure 12 (a) ΔV_{ot} and (b) ΔV_{it} induced by negative bias-temperature stressing (NBTS) with or without negative-bias irradiation as a function of temperature for Al/HfO₂ + SiO_xN_y/Si *p*MOS capacitors. The gate bias is – 0.3 V for both irradiation and NBTS. The open squares represent the degradation by NBTS alone. The half open symbols represent the sum of separate 1Mrad(SiO₂) negative-bias irradiation (Figure 9) and separate NBTS (open squares in Figure 12). The 2 Mrad(SiO₂) experiment was stopped at 150 °C because the gate leakage increased above the 100 μ C/cm² limit to avoid charging effects due to hot carriers (this current also may lead to some neutralization of ΔV_{ot}). Stress times are the same as in Figure 11. After [92].

In Fig. 13, HfO₂ capacitors are subjected to NBTS after positive bias, zero bias, and negative bias irradiation. Values of ΔV_{ot} and ΔV_{it} with and without prior irradiation are compared. Once again, combined irradiation and NBTS lead to significant enhancements of degradation over irradiation or NBTS alone. The worst case degradation here is caused by positive-bias irradiation followed by NBTS, with zero bias irradiation and NBTS causing the next worst case degradation in Fig. 13.

Fig. 14 summarizes ΔV_{ot} for the combinations of irradiation and BTS experiments performed on these HfO₂ devices; trends in ΔV_{it} are similar. The dose is 1.0 Mrad(SiO₂) in each case; the stress condition is 600 s at 75 °C. This BTS condition is not intended to predict worst-case response at end of life for devices built in these technologies; it is instead intended to represent the kind of heating that may occur during typical device operation. Of the cases in Fig. 14, the worst-case ΔV_{ot} is positive-bias irradiation followed by NBTS. This is not a realistic pair of operating conditions for a typical circuit. However, nearly as much degradation is observed for zero bias irradiation followed by NBTS. This is a typical pair of operating conditions for a pMOS transistor. Moreover, the resulting voltage shifts at ~ 1.0 Mrad(SiO₂) in Fig. 14 are large enough to jeopardize circuit response (even before interface trap effects are considered, which will add further to the shift magnitudes), even at these 2.1 nm effective-oxide thicknesses [12]. Fig. 14 shows that, for cases in which field oxide leakage in *n*MOS transistors does not dominate the circuit radiation response [131]-[133], worst-case response for these devices likely will be for pMOS devices irradiated in the "off" state, and then turned on. This contrasts significantly to typical worst-case operating conditions for MOS devices with thermal SiO2 gate oxides, in which *n*MOS transistors irradiated at positive bias usually show worst-case response [120].



Figure 13 (a) ΔV_{ot} and (b) ΔV_{it} induced by negative-bias irradiation (V_g = -0.3 V), zero-bias irradiation, and positive-bias irradiation (V_g = 0.3 V; E_{ox} = 2.4 MV/cm), followed by negative bias-temperature stressing at 1 Mrad(SiO₂) as a function of temperature for Al/HfO₂ + SiO_xN_y/Si *p*MOS capacitors. The bias applied for NBTS is - 0.3 V. The total dose is 1 Mrad(SiO₂), and the stress times are the same as Figure 11 and Figure 12. After [92].



Experiment Conditions

Figure 14 ΔV_{ot} at 1.0 Mrad(SiO₂) for Al/HfO₂+SiO_xN_y/Si *p*MOS capacitors for: 1) negative-bias irradiation and no BTS; 2) zero bias irradiation and no BTS; 3) positive-bias irradiation and no BTS; 4) no irradiation and NBTS; 5) positive-bias irradiation and PBTS; 6) negative-bias irradiation and NBTS; 7) zero-bias irradiation and NBTS; and 8) positive-bias irradiation and NBTS. The bias stressing temperature is 75 °C. Gate biases are ± 0.3 V or 0 V during irradiation, and ± 0.3 V during BTS. After [92].

The above effects should be possible to identify in lot acceptance testing if combined irradiation and BTS testing is performed. Indeed, although it was designed as a screen against interface-trap related failures in *n*MOS transistors, the "rebound" test in MIL-STD 883, Test Method 1019 (Fig. 15) may be well-suited for this purpose, since it includes irradiation to an additional 50% dose and a one-week, 100 °C biased anneal [127], [134]-[135]. Suitable bias conditions would need to be chosen to match the worst-case response, which in this case would be those in which critical *p*MOS transistors were irradiated in the "off" state, and annealed in the "on" state.



Figure 15 MIL-STD 883, Test Method 1019 flow chart. After [135].

Al₂O₃-based Devices

Next, we discuss the combined effects of irradiation and BTS on MOS capacitors with Al₂O₃ dielectrics. The Al₂O₃-based devices were subjected to negative BTS after positive, zero, and negative bias irradiation exposure to 1.0 Mrad(SiO₂). Combined irradiation and NBTS values for ΔV_{ot} and ΔV_{it} are compared. The case with the largest voltage shift (NBTS after negative bias irradiation) was determined and is

shown in Fig. 16. The pure NBTS values for ΔV_{ot} and ΔV_{it} without prior irradiation are also shown. The combined irradiation and NBTS values of ΔV_{ot} exceed the sums of the separate irradiation (Fig. 10) and BTS (Fig. 16) results at higher temperatures (> 100 °C), as illustrated by the dashed curve, in comparison to the combined stresses. For example, for 1.0 Mrad(SiO₂) irradiation in Fig. 10, $\Delta V_{ot} = -0.01$ V. After NBTS to 175 °C (Fig. 16), $\Delta V_{ot} = -0.21$ V for the 1.0 Mrad (SiO₂) case, as compared to the summed pure irradiation [1.0 Mrad (SiO₂), Fig. 10, -0.01V] and NBTS (175 °C, Fig. 16, -0.055 V) values of -0.065 V for ΔV_{ot} .

These increases likely are associated with the trapping of both positive and negative charge during irradiation, and the subsequent release of compensating electrons during the NBTS, as was the case for the HfO₂. However, the compensating electrons clearly are more stable in the Al₂O₃-based devices than the HfO₂ devices. Such variations in stability of compensating electrons have also been observed in irradiated SiO₂ [136]-[138], although not at relative densities as high as in these high- κ dielectrics. For these devices, worst-case irradiation response will be exhibited by *p*MOS transistors irradiated and annealed in the "on" condition.

In contrast to the ΔV_{ot} , the combined irradiation and NBTS values of ΔV_{it} generally are comparable (to within experimental uncertainty) with the sums of the separate irradiation (Fig. 10) and BTS (Fig.16) results. For example, for 1.0 Mrad(SiO₂) irradiation in Fig. 9, $\Delta V_{it} = -0.07$ V. After NBTS to 150 °C (Fig.16), $\Delta V_{it} = -0.096$ V for the 1.0 Mrad(SiO₂) case. This is quite similar to the summed pure irradiation [1.0 Mrad(SiO₂), Fig. 10, -0.07 V] and NBTS (150 °C, Fig.16, -0.036 V) values of -0.11 V for ΔV_{it} , as illustrated by the dashed curve.



Figure 16 (a) ΔV_{ot} and (b) ΔV_{it} induced by NBTS after negative-bias irradiation (worst case) vs. NBTS without prior irradiation as a function of temperature for Al/Al₂O₃+SiO_xN_y/Si *n*MOS capacitors. The dose is 1.0 Mrad(SiO₂). The gate bias is - 0.3 V (E_{ox}= - 2.1 MV/cm) for both irradiation and NBTS. The open squares represent the degradation by NBTS alone. The half open symbols represent the sum of separate 1 Mrad(SiO₂) negative-bias irradiation (Figure 10) and separate NBTS (open squares in Figure 16). The stress time is 600 seconds at each temperature. After [92].

That negative bias is worst case for these devices is evidence of bulk positive charge trapping in these SiO₂ films. An enhancement of oxide-trap charge away from the Si/SiO₂ interface has also been reported by Oldham et al. in commercial, non-radiation hardened oxides [139]. The apparent interface-trap buildup in these devices during negative bias may be caused by hydrogen that is released in the Si during irradiation, similar to what has been theorized during NBTS [8], [89], [94], [140], and observed for some bipolar base oxides [141].

In summary, the combined effects of irradiation and bias-temperature stress on capacitors with HfO_2 and Al_2O_3 based dielectrics stacks were investigated and compared to devices with thermal oxides. For each type of high- κ device, the combined effects of irradiation and BTS can be significantly greater than either alone, or than linear combinations of the two types of stresses. Worst-case responses in combined irradiation and BTS environments are positive (or zero) bias irradiation followed by NBTS for the HfO₂ based devices, and negative-bias irradiation followed by NBTS for the HfO₂ based devices investigated in this study. Hardness assurance tests that include irradiation and bias-temperature annealing may be used to predict the long-term radiation response and reliability of these devices, as long as worst-case conditions are properly identified and selected during device testing.

CHAPTER VII

CHARGE TRAPPING PROPERTIES DURING SWITCHED-BIAS ANNEALING

In this chapter, a series of switched-bias experiments at elevated temperatures after the irradiation was designed to investigate charge trapping characteristics during annealing. An electric field of 2 MV/cm was applied to the gate during irradiation. Alternating negative and positive bias-temperature annealing at \pm 2 MV/cm was performed at temperatures ranging from 50 °C to 150 °C. The calculated applied electric fields include the appropriate work function differences between Al and Si.

For comparison, similar switched-bias annealing was performed after constant voltage stress. In this case, an electric field of \pm 3.6 MV/cm was applied to the capacitors for 1200 seconds at room temperature. The resulting injected charge densities were ~ 10-12 mC/cm². This stress level was determined to cause equivalent damage to that produced by irradiation, to first order [57]. The contributions of oxide, interface, and border-trap charge are evaluated, and significant roles are inferred for both metastable electron traps in the near-interfacial dielectric layers and hydrogen transport and reactions at and near the Si/oxynitride interface. The results for these HfO₂ dielectrics are compared to the responses of devices with SiO₂ gate dielectrics in the literature, and physical models of charge buildup and annealing are developed to account for the results.

Post-irradiation Annealing

To explore the mechanisms that lead to the results in Figs. 11-14, alternating negative and positive BTS was applied to the worst-case device (8) in Fig. 14, starting one day after the device was irradiated and stressed at 75 °C and – 0.3 V. Values of ΔV_{ot} and ΔV_{it} are shown in Fig. 17 for PBTS and NBTS applied first for 600 s, and then for 3600 s. Values of ΔV_{ot} change more than ΔV_{it} during the switched BTS. For NBTS, ΔV_{ot} increases in magnitude; for PBTS, ΔV_{ot} decreases in magnitude.



Figure 17 ΔV_{ot} and ΔV_{it} for Al/HfO₂+SiO_xN_y/Si pMOS capacitors for a series of consecutive switched bias experiments on the same devices: 1) NBTS for 600 s after positive bias irradiation to 1.0 Mrad(SiO₂); 2) a second 600 s NBTS after a 24 hour wait while the device was stored with no bias (floating) at room temperature; 3) PBTS for 600 seconds; 4) NBTS for 600 seconds; 5) PBTS for 3600 seconds; and 6) NBTS for 3600 seconds. The gate bias is 0.3 V for all PBTS and - 0.3 V for all NBTS; the stress temperature was 75 °C. After [92].

The data of Fig. 17 are reminiscent strongly of similar bias switching experiments [52]-[53], [134], [142]-[146] for irradiated thermal SiO₂, as illustrated schematically in Fig. 18. During radiation exposure, both positive and negative charges are trapped in the bulk of the HfO₂; this is also observed in high-field stress

experiments [130]. The data of Fig. 17 show that the positive charge is trapped more stably than the negative charge, since the values of ΔV_{ot} are consistent with electrons being forced out of the oxide during the NBTS, and pulled into the oxide during the PBTS. This suggests that these electrons are located either in shallow traps in the HfO₂, at sites at the HfO₂/oxynitride interface, and/or in the oxynitride layer itself. The similarity of the responses here to those observed in SiO₂ [52]-[53], [134], [142]-[146] suggests that similar defects may be involved (e.g., O vacancies [52]-[53], [134], [142]-[146]). These results demonstrate that significant densities of dipolar defects are created during irradiation in stacks incorporating HfO₂, and that these may impact MOS postirradiation response significantly. A role for hydrogen release in the Si substrate during NBTS after radiation exposure also is likely, as discussed further below, since this process is significant for NBTI in unirradiated devices [5], [74], [88]-[89], and because bias switching effects also are observed on a smaller scale in ΔV_{it} [53], [147].



Figure 18 Schematic diagram of the gate stack used here, and electron motion during zero or positive-bias irradiation followed by positive or negative BTS. During PBTS, more electrons compensate trapped positive charge; during NBTS, fewer electrons compensate trapped positive charge. After [92].

The switched-bias annealing experiments were expanded to a fuller temperature range: 50°C ~150°C after positive bias irradiation at 1 Mrad(SiO₂), while maintaining the same stress time interval for the subsequent annealing. Fig. 19 shows results for Al/HfO₂+SiO_xN_y/Si *p*MOS capacitors. The flatband-voltage shifts due to net oxide-trap and interface-trap charge are plotted as a function of switched-bias annealing temperature. The degradation at each bias condition increases with temperature. Values of ΔV_{ot} and ΔV_{it} increase in magnitude for negative biastemperature stress (NBTS) and decrease in magnitude for positive bias-temperature stress (PBTS). A significant fraction of the reversibility in ΔV_{ot} after irradiation is similar to switched-bias experiments for irradiated thermal SiO₂ [52]-[53], [134], [136]-[138], [142]-[146], [148]. The trend is likely associated primarily with trapping of both positive and negative charge during the 10-keV X-ray irradiation, the subsequent release of compensating electrons during NBTS, and the return of these electrons into the oxide during PBTS [92]. In this case, the electrons are located either in shallow traps in the HfO₂, at sites at the HfO₂/oxynitride interface, and/or in the oxynitride layer itself, as discussed in Chapter VI, and consistent with previous work on SiO₂ [52], [134], [142]-[146], [148]. It is likely that different defects participate in the enhanced reversibility of charge trapping for these HfO₂ dielectrics than for high quality thermal SiO₂, but O vacancies almost certainly play a key role in the response of each type of device [52], [92], [134], [137], [142]-[146], [148].

The reversibility in ΔV_{it} after irradiation in Fig. 19(b) is more pronounced than is typical for similar irradiation and annealing sequences for thermal SiO₂ [149]-[150]. In particular, the response in Fig. 19(b) cannot be explained easily by the twostage buildup of interface traps associated with the release of protons in the gate dielectric and their subsequent transport under bias and reaction at the Si/dielectric interface. For this case, interface traps typically are found to build up during positive bias annealing, and stay approximately constant during negative-bias annealing [149]-[150]. However, small amounts of interface-trap buildup and reversibility of interfacetrap densities have been observed in switched-bias annealing experiments performed on SiO₂ devices at elevated temperatures [53]. This is seen for cases in which significant densities of trapped positive charge and hydrogenous species are simultaneously present in MOS devices, so these results are qualitatively consistent with the behavior of interface traps in SiO₂ under similar experimental conditions. The reversibility in ΔV_{it} (as well as a portion of the variability in ΔV_{ot}) is attributed to the motion, trapping, and reactions of protons near the Si/dielectric interface, as discussed in detail below. The increase in interface-trap density during negative-bias annealing in Fig. 19(b) is much larger (relative to the post-irradiation interface-trap density) than is commonly observed in SiO₂, which reinforces that the radiation response of MOS capacitors with high- κ dielectrics can differ significantly from the response of devices with SiO₂ gate oxides.



Figure 19 Induced ΔV_{ot} (a) and ΔV_{it} (b) for Al/HfO₂+SiO_xN_y/Si *p*MOS capacitors irradiated to 1.0 Mrad(SiO₂) with 10-keV X-rays, followed by a series of switched bias anneals at 50 to 150 °C. The gate bias for irradiation is 0.3 V. The switched bias anneals are \pm 0.3 V (PBTS, NBTS), and the stress time was 600 s each. After [151].

In addition to the reversibility of the voltage shifts in Fig. 19, an overall increase in the magnitude of the degradation is also observed with stress time, especially for higher annealing temperatures. This likely is due to additional damage associated with the increased gate currents during the elevated temperature anneal. The cumulative gate current during the entire stress period was recorded to evaluate this effect. The total injected charge densities were determined as:

$$\frac{Q}{A} = \frac{\int I(t)dt}{A} \tag{17}$$

Here A is the area of the MOS capacitor, I is the current through the oxide during the switched-bias anneals, and the integral is evaluated over the entire time for which the capacitor is under bias of a given polarity. The integral is evaluated separately for the positive and negative bias sequences.

Fig. 20 shows the injected charge densities during the post-irradiation switched-bias annealing sequences of Fig. 19. The injected charge densities increase with stress time for both NBTS and PBTS in Fig. 20. Injected charge densities increase with increasing temperature, demonstrating that the gate current is not due to pure tunnelling, but includes a more strongly temperature-dependent component. This likely is associated both with the lower barrier against electron injection into HfO₂ from Si than for SiO₂ [15], and to defects in the dielectric layers [152]. The injected charge at 150 °C is 0.9 C/cm² after four cycles of switched-bias annealing. After this sequence, the leakage current through the dielectric increases dramatically (corresponding to hard breakdown), consistent with results on thermal SiO₂ with similar defect densities [17], [57], [95], [152]. During the NBTS and PBTS cycles, the oxynitride layer is thin enough for electrons to tunnel through, enabling additional defect creation in the HfO₂, in addition to new trap creation in the interfacial oxynitride and at the interface [95], [152]. At higher temperatures, both electron

exchange and proton motion are enhanced, leading to additional defect creation. This leads to the increasing charge trapping with increasing temperature, in addition to the reversibility discussed above.



Figure 20 Cumulative injected charge densities for switched-bias annealing at elevated temperatures after irradiation. The capacitor area $A = 5.6 \times 10^{-4}$ cm². The experimental conditions are the same as Figure 18. Cumulative charge densities during PBTS are shown in the upper part of the figure, and cumulative charge densities during NBTS are shown in the lower half. These densities do not include the portions of the anneal when the device was in the opposite bias state, so the total charge is obtained by adding the charge densities in the upper and lower halves of the figure. After [151].

Post-constant voltage stress (CVS) Annealing

Fig. 21 shows results after the application of constant voltage stress (CVS) to Al/HfO₂+SiO_xN_y/Si *p*-substrate MOS capacitors. These devices were stressed with \pm 2 MV/cm at elevated temperatures up to 150 °C. Although the constant voltage stress is performed at a voltage (0.5 V) that is well below the energy needed to create electron-hole pairs in the oxide, somewhat similar trends are observed in the post-CVS defect buildup and annealing as for the case of irradiation in Fig. 19. Increasing the annealing temperature enhances the buildup of the net oxide-trap charge significantly during the initial period of negative bias annealing, but the reversibility of ΔV_{ot} after this initial increase in magnitude does not depend as strongly on temperature from 75 °C to 150 °C as was the case for the X-ray irradiations in Fig. 19. The changes in ΔV_{ot} and ΔV_{it} are primarily due to proton motion and reactions in response to the applied biases, as discussed below.



Figure 21 Induced ΔV_{ot} (a) and ΔV_{it} (b) for Al/HfO₂+SiO_xN_y/Si *p*-substrate MOS capacitors stressed at - 3.6 MV/cm for 1200 s followed by a series of switched bias experiments at 50 to 150 °C. The gate bias for constant voltage stress is – 0.5 V. The switched bias anneals are ± 0.3 V (PBTS, NBTS) and the stress time was 600 s each. After [151].

The absence of radiation-induced-hole trapping during the CVS, and the amphoteric nature of interface traps at the Si/SiO₂ interface (leading to two electrically active defect levels for each physical interface trap [48]) cause the interface trap densities to exceed the oxide-trap charge densities in Fig. 21, in contrast to Fig. 19, in which the oxide-trap charge creation dominates. This is illustrated more clearly in Fig. 22, which compares the switched-bias annealing responses at 75 °C for these high-k devices after irradiation and CVS. More oxide-trap charge than interfacetrap charge is created by radiation exposure, and more interface-trap charge than oxide-trap charge is created by CVS for these experimental conditions. Moreover, the damage levels (especially after periods of negative bias annealing) greatly exceed the damage levels immediately after irradiation, consistent with the results in ref [92]. The reversibility in ΔV_{ot} after irradiation is greater than the reversibility in ΔV_{it} ; in contrast, the reversibility in ΔV_{it} after CVS is greater than the reversibility in ΔV_{ot} . Note that nearly as much reversibility is observed in the values of ΔV_{it} after CVS as is observed in ΔV_{ot} after irradiation, illustrating that, depending on the amount and type of stress to the device, significant reversibility can be observed in either oxide or interface-trap charge in these high-k dielectrics. This contrasts with SiO₂ in which the reversibility of oxide-trap charge is typically much greater than the reversibility of interface-trap charge for irradiation or high-field stress [52]-[53], [134], [142]-[146], [148], [153].



Figure 22 Induced ΔV_{ot} (solid symbols) and ΔV_{it} (open symbols) for switched-bias annealing either after irradiation (squares) or after constant voltage stress (circles). The gate bias for irradiation is 0.3 V and the dose is 1 Mrad(SiO₂). The gate bias for constant voltage stress is – 0.5 V and the stress time is 1200 s. Both irradiation and constant voltage stress were performed at room temperature. The switched biases are ± 0.3 V at 75 °C, and the stress time was 600 s for each time interval (NBTS, PBTS) for both cases. After [151].



Figure 23 Injected charge densities for switched-bias annealing at elevated temperatures after constant voltage stress (CVS). The experimental conditions are the same as Figure 21. After [151].

Fig. 23 quantifies the increase in injected charge densities during the switchedbias annealing after CVS for the sequence in Fig. 21. Similar trends are observed to those in Fig. 20; that is, with increasing post-CVS switched-bias annealing time and/or increasing temperature, additional charge is injected into the dielectric layers, leading to the increase in defect density with increasing anneal time.

The above trends in charge trapping reversibility and defect growth are also illustrated nicely via multiple switched-bias annealing cycles. Fig. 24 displays a sequence of NBTS and PBTS cycles at 50 °C or 75 °C after constant voltage stress (CVS) to show the evolution in charge trapping with time at two temperatures that are fairly typical of device operating temperatures in practical applications. The same trends of reversibility in ΔV_{ot} and ΔV_{it} are found with increasing magnitudes of the voltage shifts as the switched-bias annealing continues. For example, at the data set for switched-bias annealing at 75 °C, the voltage shifts due to the first negative BTS are -0.06 V for ΔV_{ot} and -0.26 V for ΔV_{it} , respectively. These increase in magnitude to -0.18 V and -0.36 V for ΔV_{it} after 5 cycles of switched-bias annealing. The reversibility of ΔV_{it} is around two times larger than that of ΔV_{ot} . Moreover, ΔV_{ot} and ΔV_{it} are larger in magnitude for switched-bias annealing at 75 °C than 50 °C annealing. Strong reversibility in trapped charge densities occurs, consistent with the trends in Figs. 19 - 23. In addition, there is a general downward trend (upward in magnitude) in the voltage shifts due to new trap creation, associated with increasing cumulative charge injection during the annealing cycles.



Figure 24 Induced ΔV_{ot} (solid symbols) and ΔV_{it} (open symbols) for switched-bias annealing at 50 °C (squares) or 75 °C (circles) after constant voltage stress. The gate bias for constant voltage stress is -0.5 V and the stress time is 1200 s. Both irradiation and constant voltage stress were performed at room temperature. The switched bias anneals are ± 0.3 V at 75 °C, and the stress time was 600 s for each time interval (NBTS, PBTS) for both cases. After [151].

Physical Mechanisms for Charge Trapping Reversibility

Next, we discuss physical mechanisms for charge trapping reversibility. Figs. 25(a) and (b) illustrate schematically mechanisms that are consistent with the observed degradation in Fig. 19 and Fig. 21. The reversibility in ΔV_{ot} after irradiation is greater than the reversibility in ΔV_{it} ; in contrast, the reversibility in ΔV_{it} after CVS is greater than the reversibility in ΔV_{ot} . The enhanced reversibility in ΔV_{ot} for the post-irradiation case is consistent with large densities of both holes and electrons being trapped in the SiO₂ during irradiation [92], illustrated as mechanism (1) in Fig. 25. Such a high level of oxide-trap charge due to electron-hole pair generation and trapping in the insulator is absent in the CVS case, leading to the dominant role of hydrogen reactions, as illustrated by mechanisms (2)-(4).

Under negative bias, H^+ drift to the interface from the oxide is inhibited by the applied electric field, and Si dangling bonds are positively charged. In this case, passivation of dangling bonds by H (reaction (18)) is suppressed, since both species are of the same charge.

$$H^{+} + Si^{-} \rightarrow Si - H. \tag{18}$$

However, the depassivation of passivated dangling bonds can still occur via reaction (19) [100],

$$Si-H + H^+ \rightarrow Si^+ + H_2 \tag{19}$$

This reaction can lead to an increase of ΔV_{it} in magnitude during the negative bias anneal, if there is a source of hydrogen either directly at the interface or in the Si substrate. Possible sources of hydrogen in p-type Si are B-H complexes [74], [154] or oxygen protrusions [89]. Depassivation of a Si-H bond and the formation of an interface trap via the reaction (19) [98] are illustrated as mechanism (2) in Fig. 25 (a). A similar mechanism has been identified as a contributing factor to negative-biastemperature instability in MOS devices and for enhanced low-dose-rate sensitivity in linear bipolar transistors [140].



Figure 25 Schematic diagram of the gate stack and involved mechanisms. Electron exchange and proton (H^+) motion are depicted during negative BTS (a) or positive BTS (b). Mechanism (1) dominates during switched-bias annealing after irradiation, while mechanisms (2)-(4) dominate during post-CVS annealing. During NBTS, electrons are pulled back into the substrate after irradiation (mechanism 1(a)). H^+ can either depassivate a Si-H bond (mechanism (2a)) or be pulled off the Si-H bond when there is a spatially co-located suboxide bond (O vacancy), Hf atom, or other similar defect nearby that provides a trapping site for the proton, leaving behind a Si dangling bond that functions as an interface trap (mechanism (3a)-4(a)). During PBTS, electrons move into the oxide after irradiation (mechanism 1(b)). H^+ also is attracted to and passivates Si dangling bonds near the interface (mechanisms (2b)-(4b)). After [151].

Other mechanisms breaking Si-H bonds likely contribute to the increase in magnitude of oxide-trap charge and interface-trap charge under negative bias in Fig. 19 and Fig. 21, as also shown schematically in Fig. 25(a) (mechanisms (3) and (4)). If it is energetically favorable for the Si–H bond to break under negative bias, a proton can be released, which leads to the formation of an interface trap and an available H⁺ that can be trapped in the dielectric layer. Density-functional-theory calculations have shown that simple thermally assisted Si–H bond breaking is highly improbable for a passivated dangling bond at an otherwise defect-free (at least in the vicinity of the Si–H bond) interface under normal device operating conditions [74]. However, at simultaneously elevated temperature and a significant, applied electric field, in the presence of a nearby defect or impurity atom, it is likely that the binding energy of the H atom would be reduced.

There are at least two plausible candidates for a defect or impurity that can both facilitate the breaking of the Si–H bond and serve as a trapping site for the proton that is released in these devices. One candidate is an impurity Hf atom in the near-interfacial oxynitride (mechanism (3) in Fig. 25(a)). Density-functional-theory calculations show that isolated Hf atoms can be incorporated into the SiO₂ interlayer between the HfO₂ gate dielectric and the Si substrate during rapid thermal annealing [155]. This phenomenon was observed in scanning transmission electron microscopy studies [156] that reveal a density of Hf atoms in the SiO₂ interlayer at a high enough level (areal density of ~ 10^{13} cm⁻²) to contribute significantly to the processes in Fig. 25. Moreover, density-functional-theory based mobility calculations have found a significant degradation of mobility due to the large densities of neutral Hf atoms in the oxynitride interlayer [157]. The growth of HfO₂ via atomic layer deposition and the subsequent annealing process also can produce more suboxide bonds near the
interface with Si [158], owing to the effective "gettering" of a significant fraction of the O atoms in the oxynitride layer into the HfO₂. The Hf atoms (mechanism (3) in Fig. 25(a)) or suboxide bonds (mechanism (4) in Fig. 25(a)) in the near-interfacial oxynitride, can serve as proton traps under negative bias [140], and can release the protons to repassivate the Si dangling bonds under positive bias. That similar activation energies for BTS are observed in HfO₂ dielectrics with oxynitride interfacial layers and capacitors with thermal SiO₂ suggests that the suboxide bond mechanism may dominate this process [89], [101], [159]-[160].

Under positive bias, the decrease in ΔV_{it} is attributed to the passivation of negatively charged Si dangling bonds by protons. The protons can either be released and transport from the oxide, or be released from Hf-H or suboxide bonds, illustrated by mechanisms (2)-(4) in Fig. 25(b). In this case, there is a relatively large initial density of Si dangling bonds and a relatively high concentration of protons in the near-interfacial SiO₂, since HfO₂ is a weak diffusion barrier for hydrogenous species [161]. The transporting protons have a higher probability in this case to passivate a pre-existing defect (a negatively charged Si dangling bond) via the simple reaction (18) than to depassivate a Si-H bond and form an interface trap via reaction (19) [98], [100]. A similar mechanism has been observed to lead to a decrease in ΔV_{it} during irradiation for some high- κ dielectrics [13], emphasizing the plausibility of this mechanism in these kinds of devices. Once the defect is passivated by hydrogen, it no longer functions as an interface trap, therefore reducing ΔV_{it} in magnitude. This leads to an increase in magnitude of both ΔV_{ot} and ΔV_{it} during NBTS (more trapped protons in the oxide; more unpassivated dangling bonds), and a decrease in magnitude of ΔV_{ot} and ΔV_{it} during PBTS (fewer trapped protons in the oxide; fewer dangling bonds), consistent with the trends in the data of Fig. 19 and Fig. 21. Because there are two

defect levels for each interface trap [48], the measured value of ΔV_{it} during a C-V sweep can exceed that of ΔV_{ot} by up to a factor of two, consistent with the post-CVS results in Fig. 21. Hence, the results of Fig. 19 and Fig. 21 illustrate the competition that can occur between interface-trap buildup and passivation in devices with moderate to high interface-trap densities during irradiation and/or postirradiation annealing. The process that dominates is determined by the initial defect densities, concentrations of hydrogenous species, bias, time, dose, temperature, humidity, etc [38], [53], [99], [120], [147], [162]-[163].

Contributions from Border Traps

Because border traps can sometimes be mistaken for interface traps, I have checked the relative significance of border traps in these devices by measuring C-V hysteresis as a function of voltage ramp rate [53], [57]-[58]. The value of ΔV_{bt} is calculated from the following equations:

$$\Delta V_{bt} = \frac{\Delta N_{bt} q}{C_{ox}} \tag{20}$$

and

$$\Delta N_{bt} = \frac{\int \left| C_{reverse}(V) - C_{forward}(V) \right| dV}{qA}$$
(21)

These results are shown in Fig. 26, which were obtained after several of the irradiation and annealing sequences in Figs. 19 and 21. These border traps may be either in the HfO₂ layer or the near-interface oxynitride/Si interface. While some border traps are present, under the experimental conditions used to estimate the values of ΔV_{it} , the changes in ΔV_{it} in Figs. 19 and 21 are much larger than would be expected from at least the slower border traps that contribute to C-V hysteresis in Fig. 26.

Because similar trends in interface-trap density have been observed in previous studies of switched-bias annealing for SiO_2 -based MOS transistors in studies in which the effects of border traps and interface traps were separated using charge pumping techniques [58], it appears quite plausible that the effects observed in Figs. 19 and 21 are indeed due to interface traps.



Figure 26 ΔV_{bt} for Al/HfO₂+SiO_xN_y/Si *p*-substrate MOS capacitors after postirradiation switched-bias annealing (solid symbols) at room temperature, 75 °C, 125 °C, and 150°C for the devices of Fig. 19, at the conclusion of the measuring sequence. The open symbols represent ΔV_{bt} induced by switched bias annealing at 75 °C after constant voltage stress (same stress time as in Fig. 21). After [151].

CHAPTER VIII

CONCLUSIONS

In summary, similar activation energies for oxide-trap and interface-trap charge formation due to NBTI in MOS capacitors with SiO₂ and SiO_xN_y/HfO₂ gate dielectrics have been observed. This defect formation was originally attributed to H⁺ release from oxide protrusions in the near-interfacial Si, its diffusion along the Si/dielectric interface, followed by its reaction with Si-H to form interface traps, or its capture at a sub-oxide bond to form positive oxide-trap charge. Later theoretical calculations show the activation energy based on quasi-equilibrium reactions is in agreement with the experimental values, and a possible source of hydrogen could be from dopant sites in the depletion region of the Si substrate. These results are consistent with the key roles played by hydrogen in MOS defect formation in MOS radiation response and long-term reliability, and suggest that minimizing excess hydrogen and O vacancies may help to reduce NBTI.

Detailed studies of radiation response on capacitors with HfO_2 - and Al_2O_3 based MOS capacitors show that positive bias irradiation on HfO_2 and negative bias irradiation on Al_2O_3 dielectrics have largest voltage shifts and degradations when compared with other bias irradiations.

For each type of high- κ device: capacitors with HfO₂- and Al₂O₃- dielectrics, the combined effects of irradiation and BTS can be significantly greater than either alone, or than linear combinations of the two types of stresses. Significant densities of dipolar charge are trapped in the high- κ dielectric layers during irradiation, with the negative charge being less stable during bias-temperature stress than the positive charge. Worst-case responses in combined irradiation and BTS environments are positive (or zero) bias irradiation followed by NBTS for the HfO₂ based devices, and negative-bias irradiation followed by NBTS for the Al₂O₃ based devices investigated in this study. Hardness assurance tests that include irradiation and bias-temperature annealing may be used to predict the long-term radiation response and reliability of these devices, as long as worst-case conditions are properly identified and selected during device testing.

Detailed investigations of switched-bias annealing as a function of time and temperature after irradiation and constant voltage stress were performed on capacitors with HfO₂-based MOS capacitors. For both cases, significant reversibility of oxidetrap and interface-trap charge densities is observed. After irradiation, the dominant source of reversibility in the charge trapping is metastable electron traps associated with radiation-induced trapped positive oxide charge in the near-interfacial dielectric layers. After constant voltage stress, the motion, reactions, and trapping of protons at or near the Si/oxynitride interface are much more significant to the observed device response. Much of the correlated increases and decreases in oxide-trap charge and interface-trap charge during switched-bias annealing are attributed to protons alternatively becoming trapped at a suboxide bond or a Hf atom in the near-interfacial oxynitride layer (negative bias), and passivating a Si dangling bond at the interface (positive bias). The enhanced relative importance of proton effects during the CVS stress is a result of reduced electron-hole pair creation during low-energy constantvoltage stress, as compared to the high-energy X-ray irradiation. Additional defect growth with time was observed as a result of additional charge injection through the gate stack, during the annealing process. This defect growth increases significantly with increasing stress time and temperature. These results illustrate that positive oxide-trap charge, electron traps, and protons play significant roles in the ionizing

radiation response and long-term reliability of MOS devices with high- κ gate dielectrics. While these results are qualitatively similar to effects observed in thermal SiO₂, the magnitudes of the effects are much larger in these devices, leading to a corresponding increase in their relative significance. This work confirms that combined radiation and long-term stress environments can be quite challenging for alternative high- κ dielectrics to SiO₂, and that (absent significant processing improvements in the future) additional testing margins will be required for these kinds of devices, relative to margins used for SiO₂ gate dielectrics.

APPENDIX

A Control code # 1: For bias temperature stressing

PERL is open source software and originally developed for Unix and its variants, such as Linux. It is a stable, powerful, cross-platform programming language that does not need a compiler. Below is the script written in PERL to control the bias-temperature stressing procedure. All I need to do is run .NBTI.pl under the home/xing directory.

#!/usr/bin/perl # Input required experimental parameters \$TEMPINC=1; #get user input print "enter desired temperature (25 C to 350 C)\n"; chop(\$TEMP=<STDIN>); print "enter the reset temperature (23 C)n; chop(\$RESET=<STDIN>); print "enter a filename for the data\n"; chop(\$FILENAME=<STDIN>); print "enter a bias for the sample\n"; chop(\$BIAS=<STDIN>); print "enter a time to hold the bias on the sample for NBTI test (sec)\n"; chop(\$HOLDTIME=<STDIN>); \$I=0; \$LOOP=0; \$CYCLES=1; #here is the main loop while (\$I < \$CYCLES) { #here is the initialization of the equipment #setup the 4140 \$foo=`./volt \$BIAS`; \$foo=`./read | tail -c 10`; #setup the temp controller \$AMBIANT=`./write "#01R00"| cut -d C -f 1|tail -c 4`; \$foo=`./write \"#01M01 \$TEMP\"`; \$foo=`./write \"#01M11 360\"`; #setup the file open(OUTPUT_FILE, ">./data/\$FILENAME"); \$INITIME=`date +%s`;

```
$DISTIME=`date +%c`;
   print OUTPUT_FILE
"$FILENAME\t$DISTIME\tbias=$BIAS\ttime=$HOLDTIME (s)\n\n";
# Current monitoring and temperature control
   $ITEMP=`./write "#01R00" | cut -d C -f 1 | tail -c 4`;
   while ( $ITEMP < $TEMP )
   {
      $CURSUM=0;
      $TEMPSUM=0;
      while ( $LOOPTEMP < $ITEMP+$TEMPINC )
      {
          $LOOP++;
          $ONE=`./read`;
          ($ONE1,$ONE2)=split(/,/,$ONE);
          ($ONE3,$ONE4)=split(/I/,$ONE1);
          $ONE=$ONE4;
          $TWO=`./write "#01R00" |cut -d C -f 1|tail -c 4`;
          $CURSUM=$CURSUM+$ONE;
          $TEMPSUM=$TEMPSUM+$TWO;
          $LOOPTEMP=`./write "#01R00" |cut -d C -f 1|tail -c 4`;
      }
      $CURAVG=$CURSUM/$LOOP;
      $TEMPAVG=int($TEMPSUM/$LOOP);
      $CURTIME=`date +%s`;
      $COUNTIME=$CURTIME-$INITIME;
      $OUTTIME=`./sec_conv.pl "$COUNTIME"`;
      print OUTPUT_FILE "$OUTTIME\t$TEMPAVG\t$CURAVG\n";
     print "$OUTTIME $TEMPAVG degrees C
                                               $CURAVG amps $LOOP
samples\n";
      $CURAVG=0;
      $TEMPAVG=0;
      ONE=0;
      $TWO=0;
      $CURSUM=0;
      $TEMPSUM=0;
      $LOOPTEMP=0;
      $LOOP=0;
      $OUTTIME=0;
      $ITEMP=`./write "#01R00" | cut -d C -f 1|tail -c 4`;
   }
   $STARTHOLD=`date +%s`;
   $STOPHOLD=$STARTHOLD+$HOLDTIME;
   while ($ITIME < $STOPHOLD)
    ł
      system("sleep 5");
      $LOOP++;
      $ONE=`./read`;
      ($ONE1,$ONE2)=split(/,/,$ONE);
      ($ONE3,$ONE4)=split(/I/,$ONE1);
      $ONE=$ONE4;
      $TWO=`./write "#01R00" |cut -d C -f 1|tail -c 4`;
      $CURSUM=$CURSUM+$ONE;
      $TEMPSUM=$TEMPSUM+$TWO;
      $LOOPTEMP=`./write "#01R00" |cut -d C -f 1|tail -c 4`;
      $CURAVG=$CURSUM/$LOOP;
      $TEMPAVG=int($TEMPSUM/$LOOP);
      $CURTIME=`date +%s`;
      $COUNTIME=$CURTIME-$INITIME;
      $OUTTIME=`./sec_conv.pl "$COUNTIME"`;
```

```
print OUTPUT_FILE "$OUTTIME\t$TEMPAVG\t$CURAVG\n";
      print "$OUTTIME $TEMPAVG degrees C
                                                 $CURAVG amps $LOOP
samples\n";
      $CURAVG=0;
      $TEMPAVG=0;
      $ONE=0;
      $TWO=0;
      $CURSUM=0;
      $TEMPSUM=0;
      $LOOPTEMP=0;
      $LOOP=0;
      $OUTTIME=0;
      $ITEMP=`./write "#01R00" | cut -d C -f 1|tail -c 4`;
      $ITIME=`date +%s`;
    }
    #cool
    $foo=`./write \"#01M01 $RESET\"`;
$foo=`./write \"#01M11 020C\"`;
    while ( $ITEMP > $RESET )
    {
      $ITEMP=`./write \"#01R00\" | cut -d C -f 1|tail -c 4`;
        system("sleep 2");
    }
    #Delay at the min temp
    $foo=`./volt 0`;
    system("sleep 6");
    $I++;
    $FILENAME=$FILENAME.$I;
    system("unix2dos OUTPUT_FILE");
```

```
}
```

APPENDIX

B Control code # 2: For C-V measurements:

This program is written in the C programming language and can be used to measure

high-frequency (1 M Hz) C-V characteristics before pre and post-stress.

```
#include "ugpib.h"
#include <stdio.h>
#include <unistd.h>
#include <string.h>
#include <stdlib.h>
#define PAD4275a 3
#define SAD4275a 0
#define DELAY 100000
void send(int ud,char *command)
{
    ibwrt(ud,command,strlen(command));
}
int main(int *argc,char *argv[])
{
    float volt,vstart=-1,vstop=1,vstep=0.5;
    int fd, c;
    char setvolt[300],voltstr[300];
    int doutput[300],output[12],outstr[200];
    double delay;
    float freq;
    char setfreq[300],freqstr[300],numsampstr[300],setnumsamp[300];
    int bm,numsamp;
    int id,rd,handle,ud;
    short lnf;
    char readbuf[100000];
    char* charpoint,*endpoint;
    char trace[100];
    handle=ibfind("gpib0");
    ibpad(handle,0);
    ibrsc(handle,1);
    ibsic(handle);
    ibsre(handle,1);
    ud=ibdev(0, PAD4275a, SAD4275a, 13, 1, 0);
    vstart=(float)atof(argv[1]);
    vstop=(float)atof(argv[2]);
    vstep=(float)atof(argv[3]);
    /* start the meat of the code */
    /* initialize the measurement */
        send(ud, "*rst n");
        send(ud,"*cls\n");
      send(ud, "A2B1C3D0F17H110M2R31S0T1");/*Frequency can be changed/
```

```
/* do the cv loop */
    if(vstart > 0)
    for (volt=vstart;volt>vstop-.0001;volt=volt+vstep)
    {
      sprintf(voltstr,"%2.2f",volt);
      strcpy(setvolt,"BI");
      strcat(setvolt,voltstr);
      strcat(setvolt,"E00V\n");
      send(ud,setvolt);
      usleep(DELAY);
      bm=0;
        strcpy(output,"
                                                                ");
      strcpy(readbuf,"\0");
11
          ibrd(ud,&output,36);
      ibrd(ud,readbuf,36);
      printf("%f\t",volt);
11
      printf("%s",output);
      charpoint=readbuf;
      while((*charpoint!='+')&&(*charpoint!='-'))
          charpoint++;
      endpoint=charpoint;
      while(*endpoint!=',')
          endpoint++;
      *endpoint='\0';
      printf("%s\n", charpoint);
    }
      send(ud, "BI000E00V\n");
    }
    if (vstart < 0)
    ł
    for (volt=vstart;volt<vstop+.0001;volt=volt+vstep)</pre>
    {
      sprintf(voltstr,"%2.2f",volt);
      strcpy(setvolt,"BI");
      strcat(setvolt,voltstr);
      strcat(setvolt,"E00V\n");
      send(ud,setvolt);
      usleep(DELAY);
      bm=0;
                                                                ");
        strcpy(output,"
      strcpy(readbuf,"\0");
11
          ibrd(ud,&output,36);
      ibrd(ud,readbuf,36);
      printf("%f\t",volt);
11
      printf("%s",output);
      charpoint=readbuf;
      while((*charpoint!='+')&&(*charpoint!='-'))
          charpoint++;
      endpoint=charpoint;
      while(*endpoint!=',')
          endpoint++;
      *endpoint='\0';
      printf("%s\n",charpoint);
    }
      send(ud,"BI000E00V\n");
    }
}
```

REFERENCES

- B. E. Deal, M. Sklar, A. S. Grove, and E. H. Snow, "Characteristics of surfacestate charge (QSS) of thermally oxidized silicon," *Journal of Electrochemical Society*, vol.114, pp. 266-274, 1967.
- [2] C. E. Blat, E. H. Nicollian, E. H. Poindexter, "Mechanism of negative-biastemperature instability," *Applied Physics Letters*, vol. 69, no. 3, pp.1712-1720, Feb.1991
- [3] E. H. Nicollian, C. N. Berglund, P. F. Schmidt, and J. M. Andrew, "Electrochemical charging of thermal SiO₂ films by injected electron currents," *Journal of Applied Physics*, vol. 42, pp. 5654-5664, 1971.
- [4] S. Ogawa and N. Shiono, "Generalized diffusion-reaction model for the lowfield charge-buildup instability at the Si-SiO₂ interface," *Physical Review B*, vol. 51, no.7, pp. 4218-4230, Feb.1995.
- [5] D. K. Schroder and Jeff A. Babcock, "Negative bias temperature instability: road to cross in deep submicron silicon semiconductor manufacturing," *Journal of Applied Physics*, vol.94, no.1, pp.1513-1530, July 2003.
- [6] S. Mahapatra, P. B. Kumar, and M. A. Alam, "Investigation and modelling of interface and bulk trap generation during negative bias temperature instability of p-MOSFETs," *IEEE Transactions on Electron Devices*, vol. 51, no. 9, pp.1371-1379, Sep. 2004.
- [7] J. F. Zhang and W. Eccleston, "Positive bias temperature instability in MOSFETs," *IEEE Transactions on Electron Devices*, vol. 45, no.1, pp.116-124, Jan. 1998.
- [8] D. M. Fleetwood, X. J. Zhou, L. Tsetseris, S. T. Pantelides, and R. D. Schrimpf, "Hydrogen model for negative bias temperature instability in MOS gate dielectrics," *PV 2005-01 - ISBN 1-56677-459-4 - Silicon Nitride and Silicon Dioxide Thin Insulating Films and Other Emerging Dielectrics VIII*, edited by R. E. Sah, M. J. Deen, J. Zhang, J. Yota, and Y. Kamakura, pp. 267-278, 2005.
- [9] P. S. Winokur, "Total-dose radiation effects (from the perspective of the experimentalist)," *IEEE NSREC Short Course*, 1992.
- [10] International Technology Roadmap for Semiconductors, semiconductor industry association. Available at http://public.itrs.net[online].
- [11] R. C. Lacoe, "CMOS scaling and process history, design principles, and hardening by design methodologies," *IEEE NSREC Short Course*, 2003.
- [12] J. A. Felix, D. M. Fleetwood, R. D. Schrimpf, J. G. Hong, G. Lucovsky, J. R. Schwank, and M. R. Shaneyfelt, "Total-dose radiation response of hafnium-

silicate capacitors," *IEEE Transactions on Nuclear Science*, vol. 49, no. 6, pp. 3191-3196, Dec. 2002.

- [13] J. A. Felix, M. R. Shaneyfelt, D. M. Fleetwood, T. L. Meisenheimer, J. R. Schwank, R. D. Schrimpf, P. E. Dodd, E. P. Gusev, and C. D' Emic, "Radiation-induced charge trapping in thin Al₂O₃/SiO_xN_y/Si(100) gate dielectric stacks," *IEEE Transactions on Nuclear Science*, vol. 50, no. 6, pp. 1910-1918, Dec. 2003.
- [14] M. Houssa, G. Pourtois, M. M. Heyns and A. Stesmans, "Defect generation in high κ gate dielectric stacks under electrical stress: the impact of hydrogen," *Journal of Physics: Condensed Matter*, vol.17, pp. s2075-s2088, 2005.
- [15] E. P. Gusev, E. Cartier, D. A. Buchanan, M. Gribelyuk, M. Copel, H. Okorn-Schmidt, and C. D'Emic, "Ultrathin high-κ metal oxides on silicon: processing, characterization and integration issues," *Microelectronic Engineering*, vol. 59, no.1-4, pp. 341-349, 2001.
- [16] G. D. Wilk, R. M. Wallace, and J. M. Anthony, "High-κ gate dielectrics: current status and materials properties considerations," *Applied Physics Letters*, vol. 89, pp. 5243-5275, 2001.
- [17] G. D. Wilk, R. M. Wallace, and J. M. Anthony, "Hafnium and zirconium silicates for advanced gate dielectrics," *Journal of Applied Physics*, vol. 87, no.1, pp. 484-492, Jan. 2000.
- [18] E. P. Gusev, D. A. Buchanan, E. Cartier, A. Kumar, D. DiMaria, S. Guha, A. Callegari, S. Zafar, P. C. Jamison, D. A. Neumayer, M. Copel, M. A. Gribelyuk, H. Okorn-Schmidt, C. D'Emic, P. Kozlowski, K. Chan, N. Bojarczuk, L-A. Ragnarsson, R. Ronsheim, K. Rim, R. J. Fleming, A. Mocuta, and A. Ajmera, "Ultrathin high-κ gate stacks for advanced CMOS devices," *IEEE International Electron Devices Meeting (IEDM) Technical Digest*, pp. 451-454, 2001.
- [19] G. D. Wilk and R. M. Wallace, "Electrical properties of hafnium silicate gate dielectrics deposited directly on silicon," *Applied Physics Letters*, vol. 74, pp. 2854-2856, 1999.
- [20] L. Kang, B. H. Lee, W. J. Qi, Y. Jeon, R. Nieh, S. Gopalan, K. Onishi, and J. C. Lee, "Electrical characteristics of highly reliable ultrathin hafnium oxide gate dielectric," *IEEE Electron Device Letters*, vol. 21, pp.181-183, 2000.
- [21] B. H. Lee, L. Kang, W. J. Qi, R. Nieh, Y. Jeon, K. Onishi, and J. C. Lee, "Ultrathin hafnium oxide with low leakage and excellent reliability for alternative gate dielectric application," *IEEE International Electron Devices Meeting (IEDM) Technical Digest*, pp.133-135, 1999.
- [22] G. Lucovsky and G. B. Rayner. Jr., "Microscopic model for enhanced dielectric constants in low concentration SiO₂-rich noncrystalline Zr and Hf silicate alloys," *Applied Physics Letters*, vol. 77, pp. 2912-2914, 2000.

- [23] E. Atanassova, A. Paskaleva, R. Konakova, and V. F. Mitin, "Influence of gamma radiation on thin Ta₂O₅ structures," *Microelectronics Journal*, vol. 32, vol. 7, pp. 553-562, July 2001.
- [24] L. Manchanda, B. Busch, M. L. Green, M. Morris, R. B. van Dover, R. Kwo, and S. Aravamudhan, "High-κ gate dielectrics for the silicon industry," *IEEE IWGI*, pp. 56-60, 2001.
- [25] S. M. Sze, *Physics of Semiconductor Devices*, New York: John Wiley and Sons, 1981.
- [26] E. H. Nicollian and J. R. Brews, *MOS (Metal Oxide Semiconductor) Physics and Technology*, New York: John Wiley and Sons, 1982.
- [27] T. Ma, S. A. Campbell, R. Smith, N. Hoilien, B. He, W. L. Gladfelter, C. Hobbs, D. Buchanan, C. Taylor, M. Gribelyuk, M. Tiner, M. Copel, and J. J. Lee, "Group IV metal oxides: high permittivity gate insulators deposited from anhydrous metal nitrates," *IEEE Transactions on Electron Devices*, vol. 48, pp. 2348-2356, 2001.
- [28] S. J. Lee, H. F. Luan, C. H. Lee, T. S. Jeon, W. P. Bai, Y. Senzaki, D. Roberts, and D. L. Kwong, "Performance and reliability of ultra thin CVD HfO₂ gate dielectrics with dual poly-Si gate electrodes," *Symposium on VLSI Technology Digest*, pp. 133-134, 2001.
- [29] C. Shen, M. F. Li, H. Y. Yu, X. P. Wang, Y. C.Yeo, and D. S. H. Chan, D. L. Kwong, "Physical model for frequency-dependent dynamic charge trapping in metal-oxide-semiconductor field effect transistors with HfO₂ gate dielectric," *Applied Physics Letters.*, vol. 86, Article no. 093510, 2005.
- [30] A. Y. Kang, P. M. Lenahan, and J. F. Conley, Jr., "The radiation response of the high dielectric-constant hafnium oxide/silicon system," *IEEE Transactions on Nuclear Science*, vol. 49, no. 6, pp. 2636-2642, Dec. 2002.
- [31] M. Balog, M. Schieber, M. Michman, and S. Patai, "Chemical vapor deposition and characterization of HfO₂ films from organo-hafnium compounds," *Thin Solid Films*, vol. 41, pp. 247-259, 1977.
- [32] M. L. Green, E. P. Gusev, R. Degraeve, and E. L. Garfunkel, "Ultrathin (< 4 nm) SiO₂ and Si-O-N gate dielectric layers for silicon microelectronics: understanding the processing, structure and physical and electrical limits," *Journal of Applied Physics*, vol. 90, no. 5, pp. 2057-2121, Sept. 2001.
- [33] J. Robertson, "Band offsets of wide-band-gap oxides and implications for future electronic devices," *Journal of Vacuum Science and Technology B*, vol. 18, pp.1785-1791, 2001.
- [34] K. Ohnishi, R. N. Choi, C. S. Kang, H. J. Cho, Y. H. Kim, R. E. Nieh, J. Han, S. A. Krishnan, M. S. Akbar, and J. C. Lee, "Bias-temperature instabilities of

polysilicon gate HfO₂ MOSFETs," *IEEE Transactions on Electron Devices*, vol. 50, no. 6, pp. 1517-1524, June 2003.

- [35] K. J. Hubbard and D. G. Schlom, "Thermodynamic stability of binary oxides in contact with silicon," *Journal of Materials Research*, vol. 11, pp. 2757-2776, 1996.
- [36] J. A. Felix, M. R. Shaneyfelt, J. R. Schwank, P. E. Dodd, D. M. Fleetwood, X. J. Zhou, and E. P. Gusev, "The effects of radiation and charge trapping on the reliability of alternative gate dielectrics," in *Defects in Advanced High-κ Dielectric Nano-Electronic Semiconductor Devices*, edited by E. Gusev (Springer, Amsterdam, 2006), pp. 299-322.
- [37] F. B. McLean, H. E. Boesch Jr., P. S. Winokur, and J. M. McGarrity, "Rapid annealing and charge injection in Al₂O₃ MIS capacitors," *IEEE Transactions on Nuclear Science*, vol. NS-21, pp. 47-55, 1974.
- [38] J. A. Felix, M. R. Shaneyfelt, D. M. Fleetwood, J. R. Schwank, P. E. Dodd, E. P. Gusev, R. M. Fleming, C. D'Emic, "Charge trapping and annealing in high-κ gate dielectrics," *IEEE Transactions on Nuclear Science*, vol. 51, no. 6, pp. 3143-3149, Dec. 2004.
- [39] E. P. Gusev, C. Cabral, Jr., B. P. Linder, Y.H. Kim, K. Maitra, E. Cartier, H. Nayfeh, R. Amos, G. Biery, N. Bojarczuk, A. Callegari, R. Carruthers, S.A. Cohen, M. Copel, S. Fang, M. Frank, S. Guha, M. Gribelyuk, P. Jamison, R. Jammy, M. Ieong, J. Kedzierski, P. Kozlowski, V. Ku, D. Lacey, D. LaTulipe, V. Narayanan, H. Ng, P. Nguyen, J. Newbury, V. Paruchuri, R. Rengarajan, G. Shahidi, A. Steegen, M. Steen, S. Zafar, and Y. Zhang, "Advanced gate stacks with fully silicided (FUSI) gates and high-κ dielectrics: enhanced performance at reduced gate leakage," *IEEE International Electron Devices Meeting (IEDM) Technical Digest*, pp. 79-82, Dec. 2004.
- [40] Y. Jeon, B. H. Lee, K. Zawadzki, W. J. Qi, Y. Jeon, R. Nieh, S. Gopalan, K. Onishi, and J. C. Lee, "Effect of barrier layer on the electrical and reliability characteristics of high-κ gate dielectric films," *IEEE International Electron Devices Meeting (IEDM) Technical Digest*, pp. 797-800, 1998.
- [41] M. Copel, E. Cartier, E. P. Gusev, S. Guha, N. Bojarczuk, and M. Poppeller, "Robustness of ultrathin alminum oxide dielectrics on Si(001)," *Applied Physics Letters, vol. 78*, pp. 2670-2672, 2001.
- [42] V. Huard, M. Denais, and C. Parthasarathy, "NBTI degradation: from physical mechanisms to modelling," *Microelectronics Reliability*, vol. 46, pp. 1-23, 2006.
- [43] M. L. Green, E. P. Gusev, R. Degraeve, E. L. Garfunkel, "Ultra thin (< 4 nm) SiO₂ and Si-O-N gate dielectric layers for silicon microelectronics: understanding the processing, structure, and physical and electrical limits," *Journal of Applied Physics*, vol. 90, no. 5, pp.2057-2111, Sep. 2001.

- [44] V. Fomenko, E. P. Gusev, E. Borguet, "Optical second harmonic generation studies of ultrathin high-κ dielectric stacks," *Journal of Applied Physics*, vol. 97, article no. 083711, 2005.
- [45] F. B. McLean and T. R. Oldham, "Basic mechanisms of radiation effects in electronic materials and devices," Harry Diamond Laboratories Technical Report, no. HDL-TR-2129, Sep. 1987.
- [46] J. A. Felix, J. R. Schwank, D. M. Fleetwood, M. R. Shaneyfelt and E. P. Gusev, "Effects of radiation and charge trapping on the reliability of high-κ gate dielectrics," *Microelectronics Reliability*, vol. 44, no. 4, pp. 563-575, 2004.
- [47] J. A. Felix, H. D. Xiong, D. M. Fleetwood, E. P. Gusev, R. D. Schrimpf, A. L. Sternberg, and C. D'Emic, "Interface trapping properties of nMOSFETs with Al₂O₃/SiO_xN_y/Si(100) gate dielectric stacks after exposure to ionizing radiation," *Microelectronic Engineering*, vol. 72, no. 1-4, pp. 50-54, April 2004.
- [48] P. M. Lenahan and P. V. Dressendorfer, "Hole traps and trivalent silicon centers in metal/oxide/silicon devices," *Journal of Applied Physics*, vol. 55, no. 10, pp. 3495-3499, May 1984.
- [49] T. P. Ma, G. Scoggan, and R. Leone, "Comparison of interface-state generation by 25-keV electron beam irradiation in p-type and n-type MOS capacitors," *Applied Physics Letters*, vol. 27, no. 2, pp. 61-63, July 1975.
- [50] P. S. Winokur, J. R. Schwank, P. J. McWhorter, P. V. Dressendorfer, and D. C. Turpin, "Correlating the radiation response of MOS capacitors and transistors," *IEEE Transactions on Nuclear Science*, vol. 31, no. 6, pp. 1453-1460, Dec. 1984.
- [51] D. M. Fleetwood, "Border traps in MOS devices," *IEEE Transactions on Nuclear Science*, vol. 39, no. 2, pp. 269-271, 1992.
- [52] D. M. Fleetwood, P. S. Winokur, R. A. Reber, Jr., T. L. Meisenheimer, J. R. Schwank, M. R. Shaneyfelt, and L. C. Riewe, "Effects of oxide traps, interface traps, and 'border traps' on metal-oxide-semiconductor devices," *Journal of Applied Physics*, vol. 73, no. 10, pp. 5058-5074, May 1993.
- [53] D. M. Fleetwood, W. L. Warren, J. R. Schwank, P. S. Winokur, M. R. Shaneyfelt, and L.C. Riewe, "Effects of interface traps and border traps on MOS postirradiation annealing response," *IEEE Transactions on Nuclear Science*, vol. 42, no. 6, pp. 1698-1707, 1995.
- [54] R. E. Stahlbush, A. H. Edwards, D. L. Griscom, and B. J. Mrstik, "Postirradiation cracking of H₂ and formation of interface states in irradiated metaloxide-semiconductor field-effect transistors," *Journal of Applied Physics*, vol. 73, no. 2, pp. 658-667, 1993.

- [55] R. E. Stahlbush, E. Cartier, and D. A. Buchanan, "Anomalous positive charge formation by atomic hydrogen exposure," *Microelectronic Engineering*, vol. 28, no.1-4, pp. 15-18, 1995.
- [56] D. J. DiMaria, "Temperature dependence of trap creation in silicon dioxide," *Journal of Applied Physics*, vol. 68, no.10, pp. 5234-5246, Nov. 1990.
- [57] D. M. Fleetwood and N. S. Saks, "Oxide, interface, and border traps in thermal, N₂O, and N₂O-nitrided oxides," *Journal of Applied Physics*, vol. 79, no. 3, pp.1583-1594, 1996.
- [58] D. M. Fleetwood, M. R. Shaneyfelt, W. L. Warren, J. R. Schwank, T. L. Meisenheimer and P. S. Winokur, "Border traps: issues for MOS radiation response and long-term reliability," *Microelectronics Reliability*, vol. 35, no. 3, pp. 403-428, 1995.
- [59] S. Ogawa, M. Shimaya, and N. Shiono, "Interface-trap generation at ultrathin SiO₂ (4–6 nm)-Si interfaces during negative-bias temperature aging," *Journal of Applied Physics*, vol. 77, no. 3, pp. 1137-1148, Feb. 1995.
- [60] D. M. Fleetwood, X. J. Zhou, L. Tsetseris, S. T. Pantelides, and R. D. Schrimpf, "Hydrogen model for negative-bias temperature instabilities in MOS gate insulators," PV 2005-01 - ISBN 1-56677-459-4 - Silicon Nitride and Silicon Dioxide Thin Insulating Films and Other Emerging Dielectrics VIII, edited by R. E. Sah, M. J. Deen, J. Zhang, J. Yota, and Y. Kamakura, pp. 267-278 (2005).
- [61] E. H. Nicollian, C. N. Berglund, P. F. Schmidt, and J. M. Andrew, "Electrochemical charging of thermal SiO₂ films by injected electron currents," *Journal of Applied Physics*, vol. 42, no. 13, pp. 5654-5664, Dec. 1971.
- [62] L. Tsetseris, X. J. Zhou, D. M. Fleetwood, R. D. Schrimpf, and S. T. Pantelides, "Field-induced reactions of water molecules at Si-dielectric interfaces," *Materials Research Society Symposium Proceedings*, vol. 786, pp. 171-176, Boston, 2004.
- [63] D. M. Fleetwood, M. P. Rodgers, L. Tsetseris, X. J. Zhou, I. Batyrev, S. Wang, R. D. Schrimpf, and S. T. Pantelides, "Effects of device aging on microelectronics radiation response and reliability," *Microelectronics Reliability* (in press).
- [64] S. S. Tan, T. P. Chen, C. H. Ang, and L. Chan, "Relationship between interfacial nitrogen concentration and activation energies of fixed-charge trapping and interface state generation under bias-temperature stress condition," *Applied Physics Letters*, vol. 82, no. 2, pp. 269-271, Jan. 2003.
- [65] S. Mahapatra, P. Bharath Kumar, and M. A. Alam, "Investigation and modeling of interface and bulk trap generation during negative bias temperature instability of p-MOSFETs," *IEEE Transactions on Electron Devices*, vol. 51, no. 9, pp. 1371-1379, Sep. 2004.

- [66] S. Ogawa and N. Shiono, "Generalized diffusion-reaction models for the lowfield charge-buildup instability at the Si-SiO₂ interface," *Physical Review B*, vol. 51,no. 7, pp. 4218-4230, Feb. 1995.
- [67] Chuan-His Liu, Ming T. Lee, Chih-Yung Lin, Jenkon Chen, Y. T. Loh, Fu-Tai Liou, K. Schruefer, A. A. Katsetos, Z. Yang, N. Rovedo, T. B. Hook, C. Wann and Tze-Chiang Chen, "Mechanism of threshold voltage shift (ΔV_{th}) caused by negative bias temperature instability (NBTI) in deep submicron *p*MOSFETs," *Japanese Journal of Applied Physics*, vol. 41, pp. 2423-2425, 2002.
- [68] Y. F. Chen, et al., "Negative bias temperature instability (NBTI) in deep submicron p⁺-gate pMOSFETs," *IRW Final report*, pp. 98-101, 2000.
- [69] V. Huard, M. Denais, F. Perrier, N. Revil, C. Parthasarathy, A. Bravaix, E. Vincent, "A thorough investigation of MOSFETs NBTI degradation," *Microelectronics Reliability*, vol. 45, no. 1, pp. 83-98, Jan. 2005.
- [70] T. Yamamoto, K. Uwasawa, and T. Mogami, "Bias temperature instability in scaled p+ polysilicon gate p-MOSFET's," *IEEE Transactions on Electron Devices*, vol. 46, no. 5, pp. 921-926, May 2004.
- [71] K. O. Jeppson and C. M. Svensson, "Negative bias stress of MOS devices at high electric fields and degradation of MNOS devices," *Journal of Applied Physics*, vol. 48, no. 5, pp. 2004-2014, May 1977.
- [72] M. A. Alam and S. Mahapatra, "A comprehensive model of PMOS NBTI degradation," *Microelectronics Reliability*, vol. 45, no.1, pp. 71-81, Jan. 2005.
- [73] H. Kufluoglu and M. A. Alam, "A geometrical unification of the theories of NBTI and HCL time-exponents and its implications for ultra-scaled planar and surround-gate MOSFETs," *IEEE International Electron Devices Meeting* (*IEDM*) Technical Digest, pp. 113-116, Dec. 2004.
- [74] L. Tsetseris, X. J. Zhou, D. M. Fleetwood, R. D. Schrimpf, and S. T. Pantelides, "Physical mechanisms of negative-bias temperature instability," *Applied Physics Letters*, vol. 86, article no. 142103, Mar. 2005.
- [75] H. Aono, E. Murakami, K. Okuyama, A. Nishida, M. Minami, Y. Ooji and K. Kubota, "Modeling of NBTI saturation effects and its impact on electric field dependence of the lifetime," *Microelectronics Reliability*, vol. 45, no. 7-8, pp.1109-1114, July-Aug. 2005.
- [76] S. Gamerith and M. Polzl, "Negative bias temperature stress on low voltage pchannel DMOS transistors and the role of nitrogen," *Microelectronics Reliability*, vol. 42, pp. 1439-1443, 2002.
- [77] K. Okuyama, et al., "Water-related threshold voltage instability of polysilicon TFTs," *IEEE International Electron Devices Meeting*, (*IEDM*) Technical Digest, pp. 527-530, Dec. 1993.

- [78] M. Houssa, "Modelling negative bias temperature instabilities in advanced p-MOSFETs," *Microelectronics Reliability*, vol. 45, no. 1, pp. 3-12, Jan. 2005.
- [79] S. Tsujikawa and J. Yugami, "Positive charge generation due to species of hydrogen during NBTI phenomenon in pMOSFETs with ultra-thin SiON gate dielectrics," *Microelectronics Reliability*, vol. 45, no. 1, pp. 65-69, Jan. 2005.
- [80] S. Mahapatra and M. A. Alam, "A predictive reliability model for PMOS bias temperature degradation," *IEEE International Electron Devices Meeting (IEDM) Digest.*, pp. 505-508, Dec. 2002.
- [81] Y. Hiruta, H. Iwai, F. Matsuoka, K. Hama, K. Maeguchi, and K. Kanzaki, "Interface state generation under long-term positive-bias temperature stress for a p⁺ poly gate MOS structure," *IEEE Transactions on Electron Devices*, vol. 36, no. 9, pp. 1732-1739, Sep. 1989.
- [82] S. Maeda, et al., "Mechanism of negative-bias temperature instability in polycrystalline-silicon thin film transistors," *Journal Applied Physics*, vol. 76, no. 12, pp. 8160-8166, Dec. 1994.
- [83] M. J. Powell, C. van Berkel and J. R. Hughes, "Time and temperature dependence of instability mechanisms in amorphous silicon thin-film transistors," *Applied Physics Letters*, vol. 54, no. 14, pp. 1323-1325, April 1989.
- [84] J. M. Soon, K. P. Loh, S. S. Tan, T. P. Chen, W. Y. Teo and L. Chan, "Study of negative-bias temperature-instability-induced defects using first-principles approach," *Applied Physics Letters*, vol. 83, no. 15, pp. 3063-3065, Oct. 2003.
- [85] J. Ushio, Y. Okuyama and T. Maruzumi, "Electric-field dependence of negativebias temperature instability," *Journal of Applied Physics*, vol. 97, article no. 086101, 2005.
- [86] A. Shimizu, K. Hachimine, N. Ohki, H. Ohta, M. Koguchi, Y. Nonaka, H. Sato, and F. Ootsuka, "Local mechanical-stress control (LMC): a new technique for CMOS-performance enhancement," *IEEE International Electron Devices Meeting (IEDM) Technical Digest*, pp. 19.4.1-19.4.4, Dec. 2001.
- [87] M. M. Albert and N. H. Tolk, "Absolute total cross sections for electronstimulated desorption of hydrogen and deuterium from silicon (111) measured by second harmonic generation," *Physical Review B*, vol. 63, article no. 035308, 2001.
- [88] C. H. Lin, M. H. Lee and C. W. Liu, "Correlation between Si-H/D bond desorption and injected electron energy in metal-oxide-silicon tunnelling diodes," *Applied Physics Letters*, vol. 78, no. 5, pp. 637-639, Jan. 2001.
- [89] X. J. Zhou, L. Tsetseris, S. N. Rashkeev, D. M. Fleetwood, R. D. Schrimpf, S. T. Pantelides, J. A. Felix, E. P. Gusev, and C. D'Emic, "Negative bias temperature instabilities in metal-oxide-silicon devices with SiO₂ and SiO_xN_y/HfO₂ gate dielectrics," *Applied Physics Letters*, vol. 84, no. 33, pp. 4394-4396, May 2004.

- [90] T. Suntola and M. Simpson, *Atomic Layer Epitaxy*. New York: Chapman and Hall, 1990.
- [91] O. Sneh, R. B. Clark-Phelps, A. R. Londergan, J. Winkler and T. E. Seidel, "Thin film atomic layer deposition equipment for semiconductor processing," *Thin Solid Films*, vol. 402, no. 1-2, pp. 248-261, Jan. 2002.
- [92] X. J. Zhou, D. M. Fleetwood, J. A. Felix, E. P. Gusev, and C. D'Emic, "Biastemperature instabilities and radiation effects in MOS devices," *IEEE Transactions on Nuclear Science*, vol. 52, no. 6, pp. 2231-2238, Dec. 2005.
- [93] E. P. Gusev, E. Cartier, M. Copel, M. Gribelyuk, D. A. Buchanan, H. Okorn-Schmidt, C. D'Emic, and P. Kozlowski, "Rapid thermal and other short-time processing techniques," in Ultrathin high-к dielectrics grown by atomic layer deposition: A comparative study of ZrO₂, HfO₂, Y₂O₃ and Al₂O₃, D. L. Kwong, K. G. Reid, M. C. Ozturk, P.J. Timans, and F. Rooseboom, editors, ECS, Pennington, NJ, 2001.
- [94] A. Candelori, A. Paccagnella, M. Cammarata, G. Ghidini, and P. G. Fuochi, "Fowler-Nordheim characteristics of electron irradiated MOS capacitors," *IEEE Transactions on Nuclear Science*, vol. 45, no. 6, pp. 2383-2390, 1998.
- [95] D. J. DiMaria, E. Cartier, and D. Arnold, "Impact ionization, trap creation, degradation, and breakdown in SiO₂ films on Si," *Journal of Applied Physics*, vol. 73, pp. 3367-3384, April 1993.
- [96] M. Houssa, S. D. Gendt, J. L. Austran, G. Groeseneken, M. M. Heyns, "Role of hydrogen on negative bias temperature instability in HfO₂-based hole channel field-effect transistors," *Applied Physics Letters*, vol. 85, pp. 2101-2103, 2004.
- [97] W. L. Warren, D. M. Fleetwood, J. R. Schwank, M. R. Shaneyfelt, P. S. Winokur, R. A. B. Devine, and W. P. Maszara, "Shallow oxygen-related donors in bonded and etchback silicon on insulator structures," *Applied Physics Letters*, vol. 64, no.4, pp. 508-510, Jan. 1994.
- [98] S. N. Rashkeev, D. M. Fleetwood, R. D. Schrimpf, and S. T. Pantelides, "Proton-induced defect generation at the Si-SiO₂ interface," *IEEE Transactions* on Nuclear Science, vol. 48, no. 6, pp. 2086-2092, 2001.
- [99] S. T. Pantelides, S. N. Rashkeev, R. Buczko, D. M. Fleetwood, and R. D. Schrimpf, "Reactions of hydrogen with the Si-SiO₂ interface," *IEEE Transactions on Nuclear Science*, vol. 47, no. 6, pp. 2262-2268, Dec. 2000.
- [100] S. N. Rashkeev, D. M. Fleetwood, R. D. Schrimpf, and S. T. Pantelides, "Defect generation by hydrogen at the Si-SiO₂ interface," *Physics Review Letters*, vol. 87, no. 16, article no.165506, Oct. 2001.

- [101] S. N. Rashkeev, D. M. Fleetwood, R. D. Schrimpf, S. T. Pantelides, "Dual behavior of H⁺ at Si-SiO₂ interfaces: mobility versus trapping," *Applied Physics Letters*, vol. 81, no. 10, pp. 1839-1841, Sep. 2002.
- [102] D. M. Fleetwood, "Effects of hydrogen transport and reactions on microelectronics radiation response and reliability," *Microelectronics Reliability*, vol. 42, no. 4-5, pp. 523-541, 2002.
- [103] D. Vanderbilt, "Soft self-consistent pseudopotentials in a generalized eigenvalue formalism," *Physical Review B, Condensed matter*, vol. 41, no. 11, pp. 7892-7895, April 1990.
- [104] G. Kresse and J. Furthmuller, "Efficient iterative schemes for *ab initio* totalenergy calculations using a plane-wave basis set," *Physical Review B, Condensed matter*, vol. 54, no. 16, pp. 11169-11186, Oct. 1990.
- [105] J. H. Stathis, "Dissociation kinetics of hydrogen-passivated (100) Si/SiO₂ interface defects," *Journal of Applied Physics*, vol. 72, no. 12, pp. 6205-6207, June 1995.
- [106] K. L. Brower, "Kinetics of H₂ passivation of P_b centers at the (111) Si-SiO₂ interface," *Physical Review B*, vol. 38, pp. 9657-9666, 1988.
- [107] R. Khatri, P. Asoka-Kumar, B. Nielsen, L. O. Roellig, and K. G. Lynn, "Kinetics of hydrogen interaction with SiO₂-Si interface trap centers," *Applied Physics Letters*, vol. 65, no. 3, pp. 330-332, July 1994.
- [108] A. Stesmans, "Dissociation kinetics of hydrogen-passivated (100) P_b defects at the (111) Si-SiO₂ interface," *Physical Review B*, vol. 61, pp. 8393-8403, 2000.
- [109] L. Tsetseris and S. T. Pantelides, "Migration, incorporation and passivation reactions of molecular hydrogen at the Si-SiO₂ interface," *Physical Review B*, vol. 70, no. 15, pp. 2086-2092, 2004.
- [110] C. G. van de Walle, P. J. H. Denteneer, and S. T. Pantelides, "Theory of hydrogen diffusion and reactions in crystalline silicon," *Physical Review B*, vol. 39, pp. 10791-10808, 1989.
- [111] P. J. H. Denteneer, C. G. van de Walle, and S. T. Pantelides, "Microscopic structure of the hydrogen-phosphorus complex in crystalline silicon," *Physical Review B*, vol. 41, no. 10, pp. 3885-3888, 1990.
- [112] N. M. Johnson and C. Herring, "Diffusion of negatively charged hydrogen in silicon," *Physical Review B*, vol. 46, pp. 15554-15557, 1992.
- [113] J. Zhu, N. M. Johnson, and C. Herring, "Negative charge state of hydrogen in silicon," *Physical Review B*, vol. 41, pp. 12354-12375, 1990.

- [114] N. M. Johnson and C. Herring, "Kinetics of minority-carrier-enhanced dissociation of hydrogen-dopant complexes in semiconductors," *Physical Review B*, vol. 45, no. 19, pp. 11379-11382, 1992.
- [115] W. O. Jeppson and C. M. Svensson, "Negative bias stress of MOS devices at high electric fields and degradation of MNOS devices," *Journal of Applied Physics*, vol. 48, pp. 2004-2014, 1977.
- [116] M. L. Reed and J. Plummer, "Si-SiO₂ interface trap production by lowtemperature thermal processing," *Applied Physics Letters*, vol. 51, pp. 514-516, 1987.
- [117] R. Rizk, P. de Mierry, D. Ballutaud, and M. Aucouturier, "Hydrogen diffusion and passivation processes in *p*- and *n*-type crystalline silicon," *Physical Review B*, vol. 44, no. 12, pp. 6141-6151, 1991.
- [118] T. Zundel and J. Weber, "Dissociation energies of shallow-acceptor-hydrogen pairs in silicon," *Physical Review B*, vol. 39, no.18, pp. 13549-13552, 1989.
- [119] L. Tsetseris, X. J. Zhou, D. M. Fleetwood, R. D. Schrimpf, and S. T. Pantelides, "Dual role of fluorine at the Si-SiO₂ interface," *Applied Physics Letters*, vol. 85, no. 21, pp. 4950-4952, 2004.
- [120] D. M. Fleetwood, P. V. Dressendorfer, and D. C. Turpin, "A re-evaluation of worst-case postirradiation response for hardened MOS transistors," *IEEE Transactions on Nuclear Science*, vol. 34, no. 6, pp. 1178-1183, Dec. 1987.
- [121] D. M. Fleetwood and J. H. Scofield, "Evidence that similar point defects cause 1/*f* noise and radiation-induced-hole trapping in metal-oxide-semiconductor transistors," *Physical Review Letters*, vol. 64, pp. 579-582, Jan. 1990.
- [122] D. M. Fleetwood, S. L. Miller, R. A. Reber Jr., P. J. McWhorter, P. S. Winokur, M. R. Shaneyfelt, and J. R. Schwank, "New insights into radiation-induced oxide-trap charge through thermally stimulated-current measurement and analysis," *IEEE Transactions on Nuclear Science*, vol. 39, pp. 2192-2203, Dec. 1992.
- [123] F. B. McLean, H. E. Boesch, Jr., and T. R. Oldham, "Electron-hole generation, transport, and trapping in SiO₂," in *Ionization Effects in MOS devices and Circuits*, pp.87-192, T. P. Ma and P. V. Dressendorfer, Editors. New York: Wiley, 1989.
- [124] D. M. Fleetwood, M. J. Johnson, T. L. Meisenheimer, P. S. Winokur, W. L. Warren, and S. C. Witczak, "1/f noise, hydrogen transport, and latent interface-trap buildup in irradiated MOS devices," *IEEE Transactions on Nuclear Science*, vol. 44, no. 6, pp. 1810-1817, Dec. 1997.
- [125] F. B. McLean, "A framework for understanding radiation-induced interface states in SiO₂ MOS structures," *IEEE Transactions on Nuclear Science*, vol. 27, no. 6, pp.1651-1657, Dec. 1980.

- [126] N. S. Saks and D. B. Brown, "Interface trap formation via the two-stage H⁺ process," *IEEE Transactions on Nuclear Science*, vol. 36, no. 6, pp. 1848-1857, Dec. 1989.
- [127] D. M. Fleetwood, P. S. Winokur, and J. R. Schwank, "Using laboratory X-ray and Co-60 irradiations to predict CMOS device response in strategic and space environments," *IEEE Transactions on Nuclear Science*, vol. 35, no. 6, pp. 1497-1505, Dec. 1988.
- [128] N. S. Saks and D. B. Brown, "Observation of H⁺ motion during interface-trap formation," *IEEE Transactions on Nuclear Science*, vol. 37, no. 6, pp. 1624-1631, Dec. 1990.
- [129] P. E. Bunson, M. Di Ventra, S. T. Pantelides, D. M. Fleetwood, and R. D. Schrimpf, "Hydrogen related defects in irradiated SiO₂," *IEEE Transactions on Nuclear Science*, vol. 47, no. 6, pp. 2289-2296, Dec. 2000.
- [130] S. Zafar, A. Kumar, E. P. Gusev, and E. Cartier, "Threshold voltage instabilities in high-κ gate dielectric stacks," *IEEE Transactions on Device and Materials Reliability*, vol. 5, no. 1, pp. 45-64, Mar. 2005.
- [131] T. R. Oldham, A. J. Lelis, H. E. Boesch, Jr., J. M. Benedetto, F. B. McLean, and J. M. McGarrity, "Post-irradiation effects in field-oxide isolation structures," *IEEE Transactions on Nuclear Science*, vol. 34, no. 6, pp. 1184-1189, Dec. 1987.
- [132] J. R. Schwank, F. W. Sexton, D. M. Fleetwood, M. R. Shaneyfelt, K. L. Hughes, and M. S. Rodgers, "Strategies for lot acceptance testing using CMOS transistors and ICs," *IEEE Transactions on Nuclear Science*, vol. 36, no. 6, pp. 1971-1980, Dec. 1989.
- [133] M. R. Shaneyfelt, P. E. Dodd, B. L. Draper, and R. S. Flores, "Challenges in hardening technologies using shallow trench isolation," *IEEE Transactions on Nuclear Science*, vol. 45, no. 6, pp. 2584-2592, Dec. 1998.
- [134] J. R. Schwank, P. S. Winokur, P. J. McWhorter, F. W. Sexton, P. V. Dressendorfer, and D. C. Turpin, "Physical mechanisms contributing to device rebound," *IEEE Transactions on Nuclear Science*, vol. 31, no. 6, pp. 1434-1438, Dec. 1984.
- [135] D. M. Fleetwood and H. A. Eisen, "Total-dose radiation hardness assurance," *IEEE Transactions on Nuclear Science*, vol. 50, no. 6, pp. 552-564, Dec. 2003.
- [136] D. M. Fleetwood, P. S. Winokur, L. C. Riewe, O. Flament, P. Paillet, and J. L. Leray, "The role of electron transport and trapping in MOS total dose modelling," *IEEE Transactions on Nuclear Science*, vol. 46, no. 6, pp. 1519-1525, Dec. 1999.

- [137] D. M. Fleetwood, P. S. Winokur, L. C. Riewe, and R. A. Reber, Jr., "Bulk oxide and border traps in MOS capacitors," *Journal of Applied Physics*, vol. 84, pp. 6141-6148, 1998.
- [138] D. M. Fleetwood, P. S. Winokur, O. Flament, and J. L. Leray, "Stability of trapped electrons in SiO₂," *Applied Physics Letters*, vol. 74, no. 20, pp. 2969-2971, May 1999.
- [139] T. R. Oldham, A. J. Lelis, and F. B. McLean, "Spatial dependence of trapped holes determined from tunneling analysis and measured annealing," *IEEE Transactions on Nuclear Science*, vol. 33, no. 6, pp. 1203-1209, Dec. 1986.
- [140] L. Tsetseris, R. D. Schrimpf, D. M. Fleetwood, R. L. Pease, and S. T. Pantelides, "Common origin for enhanced low-dose-rate sensitivity and bias temperature instability under negative bias," *IEEE Transactions on Nuclear Science*, vol. 52, no. 6, pp. 2265-2271, Dec. 2005.
- [141] R. L. Pease, D. G. Platteter, G. W. Dunham, J. E. Seiler, H. J. Barnaby, R. D. Schrimpf, M. R. Shaneyfelt, M. C. Maher, and R. N. Nowlin, "Characterization of enhanced low dose rate sensitivity (ELDRS) effects using gated lateral PNP transistor structures," *IEEE Transactions on Nuclear Science*, vol. 51, no. 6, pp. 3773-3780, Dec. 2004.
- [142] A. J. Lelis, H. E. Boesch, Jr., T. R. Oldham, and F. B. McLean, "Reversibility of trapped hole annealing," *IEEE Transactions on Nuclear Science*, vol. 35, no.6, pp. 1186-1191, Dec. 1988.
- [143] A. J. Lelis, T. R. Oldham, H. E. Boesch, Jr., and F. B. McLean, "The nature of the trapped hole annealing process," *IEEE Transactions on Nuclear Science*, vol. 36, no. 6, pp. 1808-1815, Dec. 1989.
- [144] R. K. Freitag, D. B. Brown, and C. M. Dozier, "Experimental evidence of two species of radiation-induced trapped positive charge," *IEEE Transactions on Nuclear Science*, vol. 40, no. 6, pp. 1316-1322, Dec. 1993.
- [145] A. J. Lelis and T. R. Oldham, "Time dependence of switching oxide traps," *IEEE Transactions on Nuclear Science*, vol. 41, no. 6, pp. 1835-1843, Dec. 1994.
- [146] J. F. Conley, P. M. Lenahan, A. J. Lelis, and T. R. Oldham, "Electron spin resonance evidence that E'(gamma) centers can behave as switching oxide traps," *IEEE Transactions on Nuclear Science*, vol. 42, no. 6, pp. 1744-1749, Dec. 1995.
- [147] S. N. Rashkeev, D. M. Fleetwood, R. D. Schrimpf, and S. T. Pantelides, "Effects of hydrogen motion on interface trap formation and annealing," *IEEE Transactions on Nuclear Science*, vol. 51, no. 6, pp. 3158-3165, Dec. 2004.
- [148] D. M. Fleetwood, M. R. Shaneyfelt, L. C. Riewe, P. S. Winokur, and R. A. Reber, Jr., "The role of border traps in MOS high-temperature postirradiation

annealing response," *IEEE Transactions on Nuclear Science*, vol. 40, no. 6, pp. 1323-1334, Dec. 1993.

- [149] P. S. Winokur, H. E. Boesch, Jr., J. M. McGarrity, and F. B. McLean, "Twostage process for buildup of radiation-induced interface states," *Journal of Applied Physics*, vol. 50, pp. 3492-3494, 1979.
- [150] D. B. Brown and N. S. Saks, "Time dependence of radiation-induced interface-trap formation in MOS devices as a function of oxide thickness and applied field," *Journal of Applied Physics*, vol. 70, pp. 3734-3747, 1991.
- [151] X. J. Zhou, D. M. Fleetwood, L. Tsetseris, R. D. Schrimpf, and S. T. Pantelides, "Effects of switched-bias annealing on charge trapping in HfO₂ gate dielectrics," *IEEE Transactions on Nuclear Science*, vol. 53, no. 6, 2006 (in press).
- [152] D. M. Fleetwood, L. C. Riewe, P. S. Winokur, and F. W. Sexton, "Dielectric breakdown of thin oxides during ramped current-temperature stress," *IEEE Transactions on Nuclear Science*, vol. 47, no. 6, pp. 2305-2310, Dec. 2000.
- [153] Y. Nissan-Cohen, J. Shappir, and D. Frohman-Bentchkowsky, "Dynamic model of trapping-detrapping in SiO₂," *Journal of Applied Physics*, vol. 58, no. 6, pp. 2252-2261, Sept. 1985.
- [154] T. Zundel and J. Weber, "Dissociation energies of shallow-acceptor-hydrogen pairs in silicon," *Physical Review B*, vol. 39, no. 18, pp. 13549-13552, June 1989.
- [155] S. N. Rashkeev, K.van Benthem, S. T. Pantelides, and S. J. Pennycook, "Single Hf atoms inside the ultrathin SiO₂ interlayer between a HfO₂ dielectric film and the Si substrate: how do they modify the interface?" *Microelectronic Engineering*, vol. 80, pp. 416-419, 2005.
- [156] K. van Benthem, A. R. Lupini, M. Kim, H. S. Baik, S. J. Doh, J. H. Lee, M. P. Oxley, S. D. Findlay, L. J. Allen, J. T. Luck, and S. J. Pennycook, "Three-dimensional imaging of individual hafnium atoms inside a semiconductor device," *Applied Physics Letters*, vol. 87, article no. 034104, 2005.
- [157] M. H. Evans, M. Caussanel, R. D. Schrimpf, and S. T. Pantelides, "Firstprinciples modeling of double-gate UTSOI MOSFETs," *IEEE International Electron Devices Meeting (IEDM) Digest*, pp. 597-600, Dec. 2005.
- [158] M. Copel and M. C. Reuter, "Decomposition of interfacial SiO₂ during HfO₂ deposition," *Applied Physics Letters*, vol. 83, no. 16, pp. 3398-3340, Oct. 2003.
- [159] G. Lucovsky and J. C. Phillips, "Microscopic bonding and macroscopic strain relaxations at Si-SiO₂ interfaces," *Applied Physics A: Materials Science & Processing*, vol. 78, no. 4, pp. 453-459, 2004.

- [160] C. Z. Zhao, J. F. Zhang, M. B. Zahid, G. Groeseneken, R. Degraeve and S. De Gendt, "Impact of gate materials on positive charge formation in HfO₂/SiO₂ stacks," *Applied Physics Letters*, vol. 89, article no. 023507, 2006.
- [161] C. Driemeier, E. P. Gusev, and I. J. R. Baumvol, "Room temperature interactions of water vapor with HfO₂ films on Si," *Applied Physics Letters*, vol. 88, article no. 201901, 2006.
- [162] P. J. McWhorter, D. M. Fleetwood, R. A. Pastorek, and G. T. Zimmerman, "Comparison of MOS capacitor and transistor postirradiation response," *IEEE Transactions on Nuclear Science*, vol. 36, pp. 1792-1799, 1989.
- [163] I. Batyrev, M. P. Rodgers, D. M. Fleetwood, R. D. Schrimpf, and S. T. Pantelides, "Effects of water on the aging and radiation response of MOS devices," *IEEE Transactions on Nuclear Science*, vol. 53, no. 6, 2006 (in press).